

# Temi di discussione

(Working Papers)

Workforce aging, pension reforms and firm outcomes

by Francesca Carta, Francesco D'Amuri and Till von Wachter







# Temi di discussione

(Working Papers)

Workforce aging, pension reforms and firm outcomes

by Francesca Carta, Francesco D'Amuri and Till von Wachter

Number 1297 - September 2020

*The papers published in the* Temi di discussione *series describe preliminary results and are made available to the public to encourage discussion and elicit comments.* 

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

*Editorial Board:* Federico Cingano, Marianna Riggi, Monica Andini, Audinga Baltrunaite, Marco Bottone, Davide Delle Monache, Sara Formai, Francesco Franceschi, Salvatore Lo Bello, Juho Taneli Makinen, Luca Metelli, Mario Pietrunti, Marco Savegnago. *Editorial Assistants:* Alessandra Giammarco, Roberto Marano.

ISSN 1594-7939 (print) ISSN 2281-3950 (online)

Printed by the Printing and Publishing Division of the Bank of Italy

#### WORKFORCE AGING, PENSION REFORMS AND FIRM OUTCOMES

by Francesca Carta<sup>\*</sup>, Francesco D'Amuri<sup>\*</sup> and Till von Wachter<sup>+</sup>

#### Abstract

Raising statutory retirement ages has been a popular policy to increase the labor supply of older workers in the face of population aging. In this paper, we quantify the effect of a sharp and unexpected increase in retirement ages on firms' input mix and economic outcomes using Italian administrative and survey data on employment, wages, value added and capital. Exploiting information on lifetime pension contributions for the universe of employees, we are able to quantify the extra number of older workers employed by each firm as a result of the reform. We find that a 10 per cent increase in older workers implies a rise in employment of young and middle-aged workers of 1.8 per cent and 1.3 per cent, respectively. Total labor costs and value added increase broadly in line with employment, with little impact on labor productivity and unit labor costs. These results suggest older workers are valuable to employers and that pension reforms postponing retirement can remove a constraint rather than place a burden on firms.

#### JEL Classification: H55, J24, J26.

Keywords: pension reform, wages, firms and labor market outcomes.

#### Contents

1. Introduction	
2. Effects of pension reforms on firm outcomes	
3. Data	
4. The Italian pension system and the 2011 pension reform	
5. Empirical analysis	
6. Results	
7. Robustness	
8. Conclusions	
References	
Figures and tables	
Appendix	

<sup>\*</sup> Bank of Italy, Structural Economic Analysis Directorate.

<sup>\*</sup> University of California Los Angeles; NBER.

# 1 Introduction

The world is aging at a fast rate and by 2030 34 countries will be "super aged", with a share of 65+ year-olds higher than 20% of the total population (Moody's Investor Service, 2010). Germany, Japan and Italy have already met this definition.<sup>1</sup> This shift in the age structure of the population will tend to lower labor force participation and to increase the dependent population, with potential negative consequences for public pension programs and economic growth (Attanasio et al., 2007; Hansen, 1939; Storesletten, 2000; Summers, 2013).

Against this background, in the last decades governments have tried to increase the labor force participation of older individuals, often by raising the statutory retirement age (OECD, 2015). A much discussed and studied potential side-effect of these policies is that older workers may crowd out younger cohorts in the labor market (e.g., Gruber and Wise (2010); Maestas et al. (2016)). Much less is known about the impact of these policies on the economic outcomes and choices of businesses. On the one hand, larger presence of older workers may hamper firms' productivity and future growth if older workers are less innovative or less willing to take risks than younger ones (e.g., Engbom (2019)). On the other hand, older workers have substantial job experience,<sup>2</sup> and an increasing number of studies suggests that departures of senior colleagues may be detrimental to co-worker productivity (e.g., Jaeger and Heining (2020), Sauvagnat and Schivardi (Sauvagnat and Schivardi)). Moreover, the aging of the labor force has been associated with increases of productivity-enhancing automation at the industry level (Acemoglu and Restrepo (2018)), and hence firms may benefit from increased employment of experienced older workers.

This paper investigates these issues by estimating the causal effects of an exogenous increase in the share of older workers on firms' input levels, wages, value added, capital, and labor productivity. We exploit a sharp and unexpected pension reform entering into force in Italy in 2012 and use unique matched employer-employee and firm balance-sheet data to study its effects. A key feature of our data is the availability of complete pension contribution histories for all workers in our sample of firms. This allows us to calculate for each employer how many of its older employees experienced an unexpected rise in retirement age due to the unanticipated reform. To obtain complete balance sheet information and to ensure a precise measure of this shock, we focus on firms with at least 50 employees at baseline.

Based on these data, our analysis proceeds in four steps. We begin by showing that, in our setting, most older workers retire as soon as they are eligible for a public pension, such that the pension reform led to unanticipated increases in the number of older employees for

<sup>2</sup>The canonical human capital model predicts that human capital increases over the life cycle (e.g., Becker (1964); Ben-Porath (1967)), and a large number of empirical studies estimate that skills increase with age, be it general skills (e.g., Bowlus and Robinson (2012)) or skills that are specific to industry, occupation, or employer (e.g., Gathmann and Schönberg (2010); Neal (1995); Parent (2000); Poletaev and Robinson (2008); Topel (1990)).

<sup>&</sup>lt;sup>1</sup> The views expressed in the article are those of the authors only and do not involve the responsibility of the Bank of Italy. We are grateful to Massimo Anelli, Joshua Angrist, Audinga Baltrunaite, Gaetano Basso, Marco Bertoni, Giulia Bovini, Giorgio Brunello, Matteo Bugamelli, Lorenzo Cappellari, David Card, Federico Cingano, Marta De Philippis, Vincenzo Galasso, Libertad Gonzales, Simon Jaeger, Jeffrey Liebman, Erzo Luttmer, Pedro Martins, Marco Manacorda, Alexandre Mas, Claudio Michelacci, Luigi Pascali, Michele Pellizzari, Roberto Piazza, Giovanni Pica, Luigi Pistaferri, Andrea Salvatori, Benjamin Schoefer, Paolo Sestito, Alessandro Tarozzi, Eliana Viviano, and seminar and conference participants at NBER Summer Institute, NBER Longer Working Lives and Labor Demand Workshop, Stanford Working Longer and Retirement Conference, Bank of Italy, Bocconi University, University of Padova, Einaudi Institute for Economics and Finance, Brucchi-Luchino Conference, European Central Bank, European Commission (DG Employment), INPS, OECD, for very useful comments. Some of the data used in this project were provided as part of the "VisitINPS scholars" programme. We are very grateful to Massimo Antichi, Massimo Ascione, Daniele Checchi, Mariella Cozzolino, Edoardo Di Porto and Paolo Naticchioni for their help. All remaining errors are solely ours.

participating employers. We instrument the change in employment due to older workers with the unexpectedly legislated change in the share of older workers eligible to retire. To benchmark our results to the literature, we then study the impact of such unexpected rise in the share of older workers on hiring and separations of younger and middle aged workers. Our main results consist in the analysis of the effect of older workers' employment on a range of firm economic outcomes, including value added, investment, and labor costs. Finally, while our main results pertain to mid-size to larger firms in the manufacturing and services sector, we extend our analysis to the universe of firms of all sizes and in all sectors of the economy, except agriculture. For this large dataset we don't have the full information on individual working histories necessary to simulate pension eligibility, and so we use it for robustness checks only.

The paper has three key findings. First, we find that an exogenous 10% increase in the number of older workers implies a 1.8% increase in the number of young workers, and a 1.3%increase for middle aged, sustained over three years. Hence, older and younger workers seem to be complements in the firms we study. Second, while total labor costs and value added also increase, a key finding is that this occurs proportional to employment, i.e., labor costs per worker and value added per worker are unaffected. Therefore, the reform led to an expansion in output and employment in affected firms at constant average labor cost and average labor productivity. Finally, we do not detect any significant effect of an increase in the number of older workers on wages. While this may be because the Italian institutional setting features rather rigid wages, this would only enhance potential negative consequences if older workers were to represent a burden to the firm. Since our identification strategy relies on the exogenous variation of employment of older (55+) employees across otherwise identical firms, it does not allow us to detect economy-wide changes in the wage. To circumvent this problem, we analyzed the change employment and mean wages by age over time. While this shows the reform led to clear shifts in employment rates among older workers, we find no reduction in wages for those nearing retirement.

Our results provide evidence for the presence of complementarity between workers of different age classes. The fact that employment and value added increase at constant labor productivity suggests that firms are able to absorb older workers without difficulties, at least in the short term. These findings are hard to reconcile with the notion that additional older workers are a burden on firms. Instead, they appear to be more consistent with the view that older workers have skills and other attributes coveted by firms that may be difficult to replace in the labour market, as suggested by a large empirical literature on skill-accumulation in labor economics, and in line with recent related work on frictions in hiring (Jaeger and Heining, 2020). Certainly, it does not appear that an increase in older workers at the firm level is a barrier to investment or lowers labor productivity, consistent with findings in Acemoglu and Restrepo (2018) that older workers in the US appear to be complements to technologies and automation.

This paper is related to several strands of prior literature. Chiefly, our paper contributes to a small but growing number of microeconomic studies of the effect of older workers' employment on firms outcomes. A key value added of our work to this emerging literature is to study the causal effect of firm-specific quasi-experimental increases in employment of older workers on a range of firm-level economic outcomes. We thereby complement panel-based estimates that do not find a negative relationship between firm-level older workers' employment changes and firm outcomes, and cross-sectional estimates that do.<sup>3</sup>

A related macroeconomic literature studying the relationship between older workers' employment

<sup>&</sup>lt;sup>3</sup>Cross sectional analyses find that the presence of older workers tends to be negatively associated with estimates of firm productivity (Haltiwanger et al., 1999; Lallemand and Rycx, 2009; Mahlberg et al., 2009). Nevertheless, such a negative relationship is not found in studies using panel data (Daveri and Maliranta, 2007; Göbel and Zwick, 2012; Malmberg et al., 2008; Mahlberg et al., 2013) or in more structural approaches (Dostie, 2011). See Table A1 for a schematic review of the literature on the effects of aging workforce on firm outcomes.

and economic outcomes at the country, state, and industry level provides important findings complementary to ours. Accemoglu and Restrepo (2018) find that, in industries with more opportunities for automation, an increasing share of older workers leads to productivity increases. While in their model it is the relative decline in middle-aged workers that triggers automation, the increasing share of highly experienced older workers can help support this process. Engbom (2019) finds that labor force aging in the last 30 years among US states lowered worker and firm dynamism with ultimately negative effects on economic growth.<sup>4</sup> While our approach is designed to exclude confounding factors that are more difficult to control for in a macroeconomic setting, we focus on at most three years after the reform and cannot capture general equilibrium effects. Yet, our results underscore that frictions in the hiring of older workers, likely explained by availability of pension benefits and lack of worker mobility, can be costly for employers.

Our findings are consistent with a large empirical literature in labor economics that has documented increasing age-experience profiles in skills that are both general and specific to the occupation, industry, or employer. Recently, an increasing number of studies have documented the effect of a loss of co-workers, and in particular of a senior team leader, on worker productivity. Jaeger and Heining (2020), Isen (2013) and Sauvagnat and Schivardi (Sauvagnat and Schivardi) investigate the effect of death of a co-worker on worker and firm outcomes. Other papers study death of inventors (Jaravel et al., 2018), primary investigators (Azoulay et al., 2010), or the departure of professors (Waldinger, 2012). In contrast to these papers, we analyze the effect of an increase, not a reduction, of senior colleagues on worker and business level outcomes.<sup>5</sup> Our approach extends this literature since we focus on firms' outcomes. We are aided by the fact that we study the effect of a change in employment of a group of workers, not a single individual, and hence may be more able to detect firm level impacts. Other papers using worker firm-level changes in employment typically focus on worker outcomes, not firm outcomes, such as studies of the effect of job losses during group or mass layoffs. Another related paper focusing on worker outcomes is Bianchi et al. (2020), who study the effect of the same reform we analyze on careers within firm, and find that the rise in older workers leads to a reduction in wage growth for workers not in retirement age. While our methodologies differ, their results are consistent with friction preventing the hiring of older workers, such that a retirement would lead to an internal promotion. In this context, our findings suggest that these freshly promoted workers are not as valuable to the firm as the older workers they replace.

Our paper is also related to a substantial literature studying the effect of older workers' employment on younger workers' job opportunities at the macro and micro level. Studies exploiting macroeconomic variation typically do not find a negative correlation between older and younger workers' employment.<sup>6</sup> Recent microeconomic studies based on pension reforms find mixed evidence of the effect of delaying older workers retirement on younger workers' employment outcomes. For example, Martins et al. (2009) find negative effects for women but not for men in Portugal. In the Netherlands, Hut (2019) finds a negative effect of delay in retirement on younger workers employment concentrated in cash-constrained firms. Studying the same reform as we do, but focusing on smaller to mid-size employers, Boeri et al. (2017) find a negative effect of a firm's growth in the number of older workers on employment of younger workers. After an extensive robustness analysis, we conclude that the difference in results is

<sup>&</sup>lt;sup>4</sup>Feyrer (2007) uses a large panel of countries and shows that increases in the proportion of workers aged 40-49 are associated with productivity growth, while those in younger (15-39 year olds) and older (50-59 and 60+) cohorts have negative effects.

<sup>&</sup>lt;sup>5</sup>In another analysis of a positive employment shock, Doran et al. (2015) find that firms that win an H-1B visa in a lottery and hire an H-1B worker moderately reduce the employment of other workers at the firm.

<sup>&</sup>lt;sup>6</sup>Studies working at the macro level do not detect any negative trade-off between the employment rates of older and younger cohorts (Gruber and Wise, 2010; Maestas et al., 2016; Tommasino and Zizza, 2015), that - if any - is restricted to periods of economic downturns (Bertoni and Brunello, Bertoni and Brunello); some negative effects are found on young workers' occupations and wages (Mohnen, 2019).

likely to be due to the fact that our main measure of the shock relates the number of excess older workers' to total employment of the firm rather than to the pre-existing stock of older workers, a measure intended to better capture the economic impact of the reform on businesses.

The rest of the paper is organized as follows. Section 2 briefly reviews predictions from theory; Section 3 presents the data, and Section 4 describes the Italian pension reform that we exploit in the empirical analysis. Sections 5 and 6 discuss the identification strategy and present main (static and dynamic) results. Section 7 provides further robustness checks. Section 8 concludes.

# 2 Effects of pension reforms on firm outcomes

This section briefly reviews the predictions for the effect of a pension-induced reduction in the retirement rate of older workers on the main outcomes of interest: the employment of older and younger workers, wages, investment, value added, labor productivity, and profitability of firms. We begin with a frictionless, competitive, benchmark and then move to more realistic scenarios frequently considered in the context of older workers, such as rigid wages and high firing costs. We conclude that although theories based on generic production functions are very flexible, a model in which older workers have firm-specific or other hard-to-hire-for skills is a good candidate to explain our findings.

The empirical literature has found, and we confirm, that pension reforms have a strong effect on older workers' employment choices due to financial incentives and social norms, among others.<sup>7</sup> An increase in the statutory retirement age thus reduces the quit rate of older workers. In the following discussion we assume this is an exogenous shock to the firm. In the empirical analysis, we will turn to an instrumental variable strategy.

#### **Competitive Case**

Assume as a benchmark case that older workers' skills are available on the market at going wages, that there are no firm-specific skills or firing costs, and that labor markets are perfectly competitive. Hence, in every period firms can freely choose the age composition of their workforce. The production function combines capital and labor, where labor is a composite of old and young workers. In this case, in the aftermath of an increase in the statutory retirement age, wages of older workers fall and they remain fully employed. This increase in the employment of older workers affects utilization and prices of younger workers and capital depending on whether they are substitutes or complements in production with older workers. Firms' profits are unaffected since factor prices equal their marginal products. In typical cases, there will be an increase in total value added.<sup>8</sup> Finally, under standard assumptions of decreasing or constant returns to scale, average labour productivity would be expected to fall.

#### Case of Wage Rigidities and Firing Costs

In the presence of implicit or explicit long-term contracts and high seniority, often wages are deemed downwardly rigid and the actual or reputational costs of firing older employees can be high. In this scenario, raising the statutory retirement age increases the stock of older workers employed at the firm without a corresponding reduction in wages, and the firm will

<sup>&</sup>lt;sup>7</sup>E.g., Cribb et al. (2016); Lalive et al. (2017); Manoli and Weber (2016); Mastrobuoni (2009); Staubli and Zweimüller (2013).

<sup>&</sup>lt;sup>8</sup>Unless, for example, the reduction in the marginal product of other inputs offsets the increase in production from additional older workers.

optimize profits by adjusting the level of other inputs.<sup>9</sup> This adjustment again depends on the degree of complementarity of these inputs with older workers and on the diminishing marginal product of aggregate labor. When complementarities prevail over the reduction in the marginal product of labor, the firm's employment of younger workers goes up and total firm size and overall production increase.<sup>10</sup> Only if older and younger workers are perfect substitutes the firm reduces employment of younger workers and does not change the overall level of employment and production. The effect on profits is ambiguous, but is likely to be negative if firms were at an optimum before the reform.<sup>11</sup> Similarly, from an optimum, labor costs per worker would be expected to increase, and average labor productivity would be expected to fall.

#### Case of Specific Skills

It may be that older workers' skills are scarce in the labor market, for example if skills are firm-specific or if older workers are hard to replace because of frictions.<sup>12</sup> An increasing number of studies reports evidence that higher-tenured workers are indeed hard to replace (e.g., Azoulay et al. (2010); Jaeger and Heining (2020); Isen (2013); Sauvagnat and Schivardi (Sauvagnat and Schivardi); Waldinger (2012)). The retirement rate represents the depreciation rate of this stock of hard-to-replace human capital. By raising the amount of high-skilled older workers and reducing the short-run costs or constraint of obtaining these skills, an increase in the statutory retirement age thus represents a gain for the firm, rather than an inconvenience or a cost. This positive shock leads to an increase in total employment, value added, and profits. The employment of younger workers increases, whereas the response in capital depends on the degree of complementarity with the labor input. Output per worker will not decrease, and may increase depending on the degree of complementarity between workers of different age.<sup>13</sup>

These initial adjustment patterns could change in the long run. In the context of firm-specific training, consider a turnover model in which young/inexperienced workers are hired to replace old/experienced workers who retire in the next period. The steady state property of the model implies a negative trade-off between young and old workers: the number of young workers hired needs to be equal to the number of old workers who retire.<sup>14</sup> In this case, a permanent reduction in the retirement rate would tend to lower the need to hire and train younger workers. However, in the presence of high turnover costs and high quit rates, the firm may still find it

<sup>&</sup>lt;sup>9</sup>If firms face small or moderate adjustment costs, they can restore the optimum by firing older workers, leaving unchanged other input quantities with no effect on value added. If young and old workers are perfect substitutes, the firm could instead reduce hiring or increase firing of younger workers as well.

<sup>&</sup>lt;sup>10</sup>If the production function has constant returns to scale, only the presence of complementarities between young and old workers drives the effects on other inputs of an increased number of old employees.

<sup>&</sup>lt;sup>11</sup>Applying the Envelope theorem, the effect on firm's profit depends on the difference between the marginal product of older workers (given the other input quantities at the new optimal level) and older workers' wages. Diminishing marginal productivity would imply a negative effect, while complementarity with other inputs can lead to a positive effect. The presence of deferred compensation contracts for older workers would reinforce the negative effect of the pension reform on profits.

<sup>&</sup>lt;sup>12</sup>Human capital theory predicts that older workers have on average higher skills (Becker, 1962; Ben-Porath, 1967; Lazear, 2009); such result has been confirmed by a large empirical literature in labor economics analyzing general (e.g., Bowlus and Robinson (2012)), industry-specific (e.g., Neal (1995)), occupation-specific (e.g., Parent (2000); Poletaev and Robinson (2008); Gathmann and Schönberg (2010)), or firm-specific skills (e.g., Topel (1990)).

<sup>&</sup>lt;sup>13</sup>General hiring/search costs on the employer's side (recruiting, opening of a vacancy, etc.) could be present for all age classes. Retaining older, high-tenured workers would cut general recruitment costs with positive effects on profits. However, it seems implausible that this would lead to an increase in costly hiring of younger workers and an expansion in production if hiring costs had not been particularly high for older workers.

<sup>&</sup>lt;sup>14</sup>A firm could try to induce older workers to work longer through offering higher wages. Once eligible for a pension benefit, this could be very costly, due to the presence increasing marginal tax rates as pension constitutes taxable income. In steady state the cost of training a younger worker will equate the cost of retaining an older worker.

optimal to expand total employment in response to the lower retirement rate. In this case, the rise in firm's total scale can lead to a larger number of younger workers and an expansion in production over the long run as well.

# 3 Data

Analyzing the effects of an increase in the number of older workers on firm outcomes is demanding in terms of data; in particular, we need information on the age-structure of the workforce of each employer, typically not available in survey or administrative firm-level data. We also need firm-level economic outcomes, which is typically not available in worker-level data. In addition, for each employee we need their full earnings and work history to infer pension eligibility. For the purpose of our analysis, we use a unique data set that matches three different sources of data. The first one is the Bank of Italy's Survey on Industrial and Services firms (INVIND).<sup>15</sup> It is a panel of 4000 manufacturing and services firms in the private non financial sector with 20+ employees (representative of 70% of total sales in the Italian economy). The second source of data is the Social Security administrative data set, provided by INPS – the Italian National Social Security Institute –, with full working histories, wages and main job-related socio-demographic characteristics of all workers employed at least one day at INVIND firms during the 2005-2015 time interval. In particular, all the needed information is available in order to retrieve the exact year in which the individual is eligible for a public pension (that is, gender, age and years of paid social security contributions); for a subset of workers, it is also possible to observe when the individual actually claims the public pension.<sup>16</sup> Third, we use CEBI (Centrale dei Bilanci), which contains full balance sheet information for each firm in every year (total labor costs and valued added, capital).

We restrict the analysis to the years 2010-2014. The far reaching pension reform used as an exogenous shifter for older workers' labor supply entered into force in January 2012. We are able to follow our firms for two years before and two years after the reform. Another pension reform took place in 2008 that could potentially confound the results, while a major labor market reform, coupled with hiring incentives, was legislated in 2015 (Sestito and Viviano, 2018).

We restrict the analysis to firms that employed at least 50 employees in the year they first appeared in the sample; we implement this restriction for two reasons: i) balance sheet information can be patchy and variables can show very volatile dynamics in smaller firms, ii) since the INVIND survey covers 20+ firms only, there could be non random attrition of shrinking firms as they cross the 20 employees threshold from above and of course such phenomenon is more likely to happen in smaller firms (D'Aurizio and Papadia, 2019). Anyway, we statistically test whether the reform we study had an impact on firm exit and entry into the sample. Finally, we restrict the data to firms sampled in each of the five years and do not have missing values on variables such as capital, labor costs and value added. Our final sample is a balanced panel of 1,025 firms/year for a total of 5,125 observations. Descriptive statistics for the main variables are reported in Table A2.<sup>17</sup>

In a robustness check, we also replicate the main analysis on a data set covering the whole population of Italian firms who paid social security contributions for at least one day in a given year between 2010 and 2014 (1.5 million firms). For this large matched employer-employee dataset we do not have the information needed to calculate the minimum age at which individuals

<sup>&</sup>lt;sup>15</sup>See Bank of Italy (2014) for a description of this data source.

<sup>&</sup>lt;sup>16</sup>This information is only available for workers born in 24 birth dates a year.

<sup>&</sup>lt;sup>17</sup>All estimates presented in the paper are unweighted, but the inclusion of survey weights would not alter the results in any way. We also carried out the analysis on a balanced or unbalanced sample for most outcomes and got similar results. Results for the weighted regressions are not reported here but are available upon request.

can claim a public pension. For this reason and the reasons mentioned above we use it mainly as a robustness check.

# 4 The Italian pension system and the 2011 reform

In the last decades the Italian pension system has been revised through a long term reform process aimed at improving its financial sustainability. In our empirical analysis we take advantage of the pension reform legislated at the end of 2011 and that unexpectedly increased the minimum legal requirements for public pension eligibility (Law 214/2011, known as "Fornero Reform"). The reform passed at the end of 2011 (December 22) during the sovereign debt crisis, and was effective from January 1, 2012. The reform was completely unexpected, as confirmed by the incidence over time of google searches for "pension reform" that had an all-time peak just in December 2011 (Figure A1 of the Appendix).

As in many OECD countries, the Italian pension system is characterized by a large first pillar consisting of public pension funds and by almost negligible second and third pillars (respectively, compulsory and voluntary<sup>18</sup> pension funds). Two types of work-related pension benefits are available and give access to full retirement: old age and seniority schemes. Eligibility for the first one is mainly based on workers' age. For the second, it mainly depends on the number of accrued years of social security contribution.

The substantial changes in the eligibility rules introduced by the reform for both the *seniority* and the *old age* pensions are summarized in Figure 1. Such changes implied an overnight increase in the average Minimum Retirement Age (MRA) of about 3 years for individuals aged 55 and more. We provide full details of these changes in Section A.1 of the Appendix. The new rules in place since 2012 allowed workers who were already eligible for a public pension when the bill passed to retire under the pre-reform rules, without losing their eligibility.

#### 4.1 Simulation of pension eligibility and Minimum Retirement Age

The sudden pension reform led to unexpected changes in public pension eligibility and a reduction in the retirement rate. Workers were affected differently depending on their age, gender and years of accrued social security contributions. The policy change also increased the number of older workers more in some firms than others, partly because of differences in the number of older workers near retirement age. In addition, given eligibility can also derive from seniority and gender, firms experienced differential increases in older workers' employment even for a given age structure.

In order to reconstruct the share of eligible workers at the firm level and its unexpected changes over time we follow several steps. First, we need to recover the minimum age at which an individual has access to pension benefits. We call it Minimum Retirement Age (MRA) – the minimum between the retirement age for old age and seniority pension (see Section 4 and Figure 2, bottom panel) – and it is computed for all individuals in each year, on the basis of three characteristics (age, gender, number of accrued years of social security contribution) and according to two different sets of pension rules. First, the MRA is computed according

<sup>&</sup>lt;sup>18</sup>The legislative decree n. 252/2005, implemented in 2007, introduced an automatic enrolment mechanism for voluntary pension funds: if an employee does not make an active choice after a six-month period (counting from January 1, 2007 for old employees and from the hiring date for new employees), the severance payment will automatically be assigned to an occupational pension plan (typically, the industry-wide occupational plan). However, according to Covip (2018), in 2017 less than 30% of the Italian working population has signed a contract with a private pension fund; however, private pension benefits are conditional on the eligibility for a public pension.

to pension rules in place at time t ( $MRA_{it}|Law_{t|t}$ ); second, it is determined on the basis of pension rules known at t-1 for time t ( $MRA_{it}|Law_{t|t-1}$ ). Since the MRA is determined also by the number of accrued years of social security contribution at the end of the individual working career, we assume that individuals in our sample will accumulate years of contribution continuously from the year of observation onward.

As a second step, we define workers' pension eligibility at t according to the pension rules in place at time t, and the expected eligibility at time t based on the rules in place at time t - 1:

$$ELIG_{it}|Law_{t|t} = \begin{cases} 1 & if \ age_{it} \ge MRA_{it}|Law_{t|t}, \\ 0 & otherwise. \end{cases}$$
(1)

and

$$ELIG_{it}|Law_{t|t-1} = \begin{cases} 1 & if \ age_{it} \ge MRA_{it}|Law_{t|t-1}, \\ 0 & otherwise. \end{cases}$$
(2)

The dummies  $ELIG_{it}|Law_{t|t}$  and  $ELIG_{it}|Law_{t|t-1}$  refer, for individual *i*, to eligibility in *t* according to the pension rules in place at time *t* and t-1, respectively.

Once we have reconstructed actual and expected eligibility and the MRA at the individual level, we can assess the magnitude of the changes that were introduced by the 2011 pension reform at the aggregate level and across firms.

Based on these definitions, our calculations show that the law had large effects on the average retirement age. In the top panel of Figure 2, the dashed line identifies for each year the share on total private sector employees of workers that in year t were expected to become eligible to retire according to the law in place in year t - 1, while the solid one refers to the share of individuals who actually became eligible in year t with the legislation in place at the beginning of year t. Any divergence between the two is due to unexpected changes in eligibility criteria between t - 1 and t. From the graph it is clear that no unexpected changes take place in any year but in 2012 – the first year of implementation of the reform. In that year, the share of workers who were expected to become eligible to retire was equal to 2.0%, and it actually went down to 1.2%. In Figure 2, we also show the expected MRA at t given the law in place in t - 1 and the actual simulated MRA for workers aged 55 or more; also in this case the pattern is the same, apart from the large unexpected increase in the MRA that takes place in 2012, equal on average to 3 additional years in order to reach eligibility.

Looking at the aggregate level, such changes seem to have had a clear impact on the share of 55+ workers employed in our sample of firms. In Figure 3, we show in dark grey the actual shares of 55+ workers on total employment in our sample, and in light grey the projected evolution of such share based on the pre-2012 trend. While the data points overlap in 2010 and 2011, a wedge opens in 2012 and increases over time. In 2014, the actual share of 55+ workers is slightly below 15 p.p., almost 3 p.p. higher than the level projected according to the pre-2012 trend.

# 5 Empirical analysis

#### 5.1 First differences OLS

The goal of the analysis is to measure the causal effect of an increase in the employment of older workers on firm outcomes. We start by exploiting the panel structure of our data and estimating the following specification in first differences:

$$\Delta y_{j,t} = \alpha + \beta \left( \frac{OLD_{j,t} - OLD_{j,t-1}}{EMPL_{j,t-1}} \right) + \phi X_{j,2010} + d_{st} + \epsilon_{j,t}$$
(3)

where j and t respectively identify firm and year and  $y_{j,t}$  is a set of firm outcomes;  $\frac{OLD_{j,t}-OLD_{j,t-1}}{EMPL_{j,t-1}}$ is the change in older workers employment (55+ year-old) as a share of firm total employment in t-1;  $d_{st}$  are sector (manufacturing or services) by year fixed effects and  $X_{j,2010}$  are the shares of middle aged (35-54) workers and old (55+) workers on total employment in the initial year of the analysis (2010).

We look at a range of different outcomes at the firm level: (a) net change in the number of workers aged 15-34 and 35-54 (equal to total hiring minus total separations); (b) gross changes in hiring and separations separately by temporary and permanent contracts (c) full time equivalent wage for 15-34, 35-54 and 55+ age classes (d) firm total labor costs, value added, capital, both total and in per worker terms. For changes in employment (a), the dependent variable  $\Delta y_{i,t}$ is defined as  $\frac{y_{j,t}-y_{j,t-1}}{EMPL_{j,t-1}}$ ; for variables in (b) and (c), the dependent variable  $\Delta y_{j,t}$  is defined as  $y_{j,t} - y_{j,t-1}$ 

 $y_{j,t-1}$ 

We choose to carry out all our estimates in first-differences since in this way we control for time-invariant unobservable characteristics. To also control for firm specific trends, in our main results we also add firm fixed effects  $f_j$  to the first differences estimates. Year fixed effects absorb a common non-linear trend.

Even though they control for time invariant unobserved heterogeneity and possible linear firm-specific trends, estimates based on equation 3 may still be biased if net employmentvariation of older workers is endogenous to firms' demand conditions. First, one would expect that older workers more likely keep on working rather than retire if employed in a booming firm to enjoy higher future wages. Second, booming firms might retain more workers of any age class, invest more, and produce more. This would lead  $\beta$  in equation 3 to be upward biased reflecting, for example, a spurious positive correlation between a rise in older workers' employment and a rise in value added.

In our empirical analysis, we exploit the panel nature of our data to directly assess whether there are pre-existing trends that are correlated with the change in the fraction of older workers around the reform (Section 7.1). Since we do not find evidence of such trends, we believe our OLS models in itself are informative. In addition, in our main estimation strategy we use the exogenous variation introduced by the reform to implement instrumental variable estimates.

#### 5.2First differences IV

The unexpected change in the minimum retirement age allows us to overcome these endogeneity issues by providing an exogenous shifter in the supply of older workers. This is because in Italy, during this period, pension wealth is maximized when claiming the public pension as soon as eligibility is reached. Hence, most workers claim at the MRA: unexpected changes in the pension rules thus provide exogenous variation in older workers employment in otherwise identical firms.

We test the validity of these assumptions by estimating the probability to claim pension benefits – and of being employed – as a function of the number of years to or from the time in which the individual actually reaches pension eligibility.

We estimate the following individual-level event-study model:

$$Y_{it} = \alpha + \sum_{k \ge -m}^{M} \phi^k D_{it}^k + \psi X_{it} + \epsilon_i$$
(4)

where Y is equal to one if the individual i claims a public pension (is employed) at time t<sup>19</sup> X are individual-level characteristics (age, gender, years of social security contributions),

 $<sup>^{19}\</sup>mathrm{Here},$  we adopt the loosest possible definition of employment, defining individuals as employed in year t if

 $D_{it}^k$  is an indicator for the k-th period before or after eligibility is reached (from *m* periods before to *M* periods after eligibility onset (k = 0)). The coefficient  $\phi^k$  captures the change in claiming/employment rate for each *k*.

In Figure 4 we show the point estimates for  $\phi^k$  from regression 4. The clear discontinuity in the year in which eligibility is reached shows that most individuals claim a public pension as soon as they are entitled to it (top panel) and stop working (bottom panel). Indeed, there is an almost deterministic link between pension eligibility and retirement (confirming the results of Battistin et al. (2009)).

This result is not surprising since for all workers who started working before 1993 (including those directly affected by the reform we study) the public pension is a Defined Benefit plan; the transfer a retiree receives is an approximately fixed percentage of the last ten years' average wage. Once eligibility is reached, postponing the claim results in a negligible increase in the amount of the transfer received (through seniority related wage growth) that does not make up for the forgone transfers. The private pension system plays a minimal role.<sup>20</sup>

#### 5.2.1 First stage

We use the exogenous variation in MRA provided by the reform to implement an instrumental variable (IV) strategy. We use the increase in the share of older workers that was unexpected by the firm as an instrument for the potentially endogenous change in the share of older workers in equation 3. To do so, we need to simulate eligibility for pension benefits for all older workers in a given firm, something that is typically hard to do since it requires knowledge of complete working histories in order to reconstruct the number of years of social security contributions. However, our data allow us to simulate eligibility under the old and the new law.

Once eligibility is simulated at the individual level, we calculate the share of eligible workers losing pension eligibility at the firm level. We focus on year 2012, since it is the only year in which the pension rules changed unexpectedly in the years we study. We show in Figure 5 the distribution of the share of a firm's total workforce of 55+ workers losing eligibility because of the reform we study. In other words, we plot the distribution of our instrumental variable

$$z_{jt} = \frac{ELIG_{j,2012} | Law_{2011} - ELIG_{j,2012} | Law_{2012}}{EMPL_{j,2011}}.$$

The instrument displays a large amount of variation across firms. A non-negligible fraction of firms is unaffected by the policy change (with share zero). Among firms experiencing a reduction in the fraction of workers eligible to retire, the reduction in the fraction eligible is 3 per cent among the total workforce or less.

To get a sense of the variation implied by the instrument, it is useful to decompose it into the share of workers age 55+ employed at the firm in 2011 that lose eligibility, and the share of workers 55+ among a firms' total employment in 2011. Figure 3 shows that the average share of older workers in 2011, the year prior to the reform, was about 10.3%. Hence, for example a 2% (1%) increase in the share among *total employment* for the average firm would imply roughly 20% (10%) of workers age 55+ had their MRA increased by the reform.

Turning to the first stage, we estimate the following regression for years 2011-2012:

$$\frac{OLD_{j,2012} - OLD_{j,2011}}{EMPL_{j,2011}} = \delta + \beta \frac{ELIG_{j,2012} | Law_{2011} - ELIG_{j,2012} | Law_{2012}}{EMPL_{j,2011}}$$

they had worked at least for one day in that year.

 $<sup>^{20}</sup>$ In 2015 24.2% of workers were enrolled in private pension funds; private pension benefits are usually small compared to the public ones.

$$+\gamma \frac{ELIG_{j,2012}|Law_{2011}}{EMPL_{j,2011}} + X_{j,2010} + d_s + \epsilon_{j,2012}$$
(5)

where  $\frac{ELIG_{j,2012}|Law_y}{EMPL_{j,2011}}$  is the fraction of eligible workers in 2012 according to the law in place in year t with t = 2011, 2012. We add sector fixed effects  $d_s$  and the share of 55+ and 35-54 y.o. workers in 2010 ( $X_{j,2010}$ ) as control variables.

The exclusion restriction implies that, conditional on the control variables, the change in the share of workers who are eligible to retire in 2012, as determined by the unexpected change in the law taking place between 2011 and 2012,<sup>21</sup> changes the fraction of older workers employed at the firm but is not correlated with unobserved firm-level demand shocks. The key identifying assumption is supported by the following: i eligibility is determined by the interaction between pre-determined characteristics of the workers (some of them difficult to observe with precision by firms, such as years of paid social security contributions) and rules that changed significantly and in unexpected ways (see Section 4 and Tables A3 and A4 in the Appendix), ii individuals in our sample retire as soon as they reach eligibility, given the incentives provided by the institutional setting (as discussed in Section 4 and shown in Figure 4).

Table 1 reports the results of our first-stage regression; the main coefficient has the right sign: at the firm level, a one percentage point unexptected decrease in the share of eligible workers is associated to a 0.56 percentage point increase in total employment due to 55+ workers (the estimate is statistically significant at the 1% level). The F-test of excluded instruments equal to 32 signals the instrument has power.

We conduct two additional tests in order to assess the validity of our instrument. First, in Figure A2 of the appendix we report on the X axis - for each quintile of the respective distribution - the averages of the residuals of a regression of the instrument on the controls, and on the Y axis the corresponding averages for a regression of the instrumented variable on the controls. The relationship between the instrumented variable and the instrument is monotonic and quite stable across the firm-level distribution of changes in eligibility. Second, we run a set of placebo estimates in which we regress the instrumented 55+ employment change taking place in 2012 on the cumulative changes of actual 55+ employment taking place between 2010 and 2014. Figure 6 shows that in the two years preceding the reform (2010 and 2011), changes in 55+ employment at the firm level were uncorrelated with the instrumented 55+ employment change taking place in 2012. After 2012, estimated coefficient values remain around one, signalling that the increase in 55+ employment taking place when the reform entered into effect was still visible two years after its inception. This is consistent with the fact that the average delay in MRA implied by the reform for 55+ workers was equal to three years (e.g., see the second panel of Figure 2).

### 6 Results

Throughout our main empirical analysis, we discuss both our OLS estimates based on panel data and the IV estimates. While the IV results are our preferred estimates, we view the OLS estimates as providing corroborating information for the following reasons: we find no evidence of firm-specific trends either when including firm fixed effects or in our dynamic analysis (Subsection 7.1); in many cases our OLS and IV estimates are qualitatively very similar and are not statistically different from each other; the IV estimates tend to have a higher variance since by design they use much less variation than the panel data.

 $<sup>^{21}</sup>$ See Table A3 and the appendix for details of changes in pension legislation.

#### 6.1 Employment and wages

We find that an exogenous increase in employment of older workers at the firm leads to an increase in employment in other age classes as well. This occurs through an increase in hiring and a (somewhat smaller) decrease in separations. The net increase in employment is driven by increases in both fixed-term and permanent positions for younger (15-34) workers and by a rise in permanent positions for the middle-aged (35-54). Finally, we find no clear impact on wages for workers of any age.

We start by looking at the results for *net employment* (Table 2). We find a strong and positive association between variation in 55+ employment and employment of young (15-34 years old) and middle-aged (35-54 years old) workers. These results hold when estimating equation 3 both without and with firm fixed effects, and are nearly identical across these two specifications; results are also very similar when looking at the balanced or unbalanced panel (Column 1-2 and 3-4, respectively), signalling our findings are not affected by sample selection over time.<sup>22</sup>

We also restrict the analysis only to employment changes taking place between the reform year (2012) and the previous one, which allows us to directly compare OLS with IV estimates (bottom panel of Table 2). For 15-34 workers the coefficient estimate remains very similar in 2011-12 for both OLS and IV estimates; for relatively more mature (35-54) workers, point estimates do not move when looking at the single-year first difference OLS, but decrease by a half when considering the corresponding IV results.<sup>23</sup>

Taking the IV estimates as our favorite coefficient estimates, a 1 percentage point increase in total employment due to 55+ workers would imply a 0.5 percentage point increase in the share of younger workers. The implied elasticities between old and youth (middle aged) employment are equal to 0.018 (and 0.013), meaning that a 10% increase in the number of older workers due to the reform would imply a 1.8% (1.3%) increase in the number of young (middle aged) workers.<sup>24</sup>

Tables 3 and 4 analyze more in detail the dynamics behind the net employment changes we find in terms of *hiring and separation margins* and changes in *permanent and fixed term contracts.* The positive employment variation for 15-34 workers is mostly attributable to an increase in the hiring rate (Column 1, Table 3), while separations move little (Column 3). Looking at 35-54 workers, we find both an increase in hiring and a decrease in separations; results are qualitatively similar across specifications, but the IV estimates are not statistically different from zero.

We also find that the contribution of temporary and permanent contracts in explaining the net employment increases differ by age (Table 4). Overall, for 15-34 individuals a substantial share of the increase in net employment occurs via a rise in fixed-term contracts; in contrast, in response to a rise in the share of older workers, workers age 35-54 are substantially more likely to be hired under a permanent contract. The magnitudes of the point estimates of the OLS specifications and IV model vary somewhat, especially for the middle aged, but the qualitative findings are similar across specifications.

 $<sup>^{22}</sup>$ The positive relationship between firm level-employment changes among 55+ aged workers and of workers aged 15-34 is also apparent in non-parametric plots. We first separately regressed net employment changes among 55+ and 15-34 y.o. on controls. Figure A3 of the Appendix shows, by quintiles of the residuals of 55+ variations, a connected scatterplot of the mean of 55+ and 15-34 variation residuals. A positive monotonic relationship is apparent.

<sup>&</sup>lt;sup>23</sup>By the formula of a standard Hausman test,  $t = (b^{ols} - b^{iv})/\sqrt{\sigma_{iv}^2 - \sigma_{ols}^2}$ , the difference in the estimates is statistically significant.

 $<sup>^{24}</sup>$ The elasticities are obtained by multiplying the coefficient estimate by the inverse of the pre-existing share of the specific age class in total employment. As we saw in Subsection 5.2.1, at the average share of older workers in employment, a 10% increase in the number of older workers would correspond to a rise in 1% of the share of older workers in employment due to the pension reform.

Moving on to wages, Table 5 shows the impact of an increase in the share of workers age 55+ on year-on-year changes in Full Time Equivalent (FTE) daily wages.<sup>25</sup> We find that employment growth in the 55+ age class is associated with a decrease in FTE wage growth concentrated in their own age class that is broadly similar across specifications (Column 3). The effect is not statistically different from zero for the IV estimate, our preferred specification. Point estimates are very close to zero and not statistically significant in the other age classes for all specifications.<sup>26</sup>

Increases in employment coupled with no or small wage reductions can be rationalized by the fact that, in the Italian institutional setting, wages tend to be rigid in the short run due to the importance of collectively bargained national contracts setting wages for three years in advance (see Adamopoulou et al. (2016)); moreover, should an increase in employment for any of the three age classes imply an increase in their wages in the whole economy, we would still fail to see them, given the fact that our identification strategy is based on variation across firms and can only spot differential changes in wages at the firm level. Nevertheless, when looking at the wage-age gradient for the economy as a whole, we find little evidence for relative wage adjustment (Figure A21 of the appendix), and if anything an increase in mature workers' wage, that could also be due to their changing composition due to the pension reform.

To directly assess this possibility, we exploited the fact that our reform provides variation in pension eligibility across age groups. The average rate of exit from employment by age among all workers in our sample in the years before (2009-2011) and after the reform (2012-2014) is shown in Figure A4 separately for men and women. While - due to the complex nature of pension eligibility - no single age is affected, one sees a clear increase in the age of exit from employment. In contrast to the large changes in employment, when we compare average FTE daily wages in the years before (2009-2011) and after the reform year (2012-2014), they barely changed for workers 50-59 (Figure 7). Within each period we see a rapid decline in wages as workers retire, most likely due to selection. The timing of this decline shifts with age, consistent with the increased retirement age. As a result, if at all wages tend to increase for workers in the range of 60 to 65 years range. However, given strong potential changes in the sample composition across ages as workers retire, we do not interpret this as causal.

#### 6.2 Firm outcomes

A key advantage of our setting is that it allows us to study firms' economic responses to a rise in employment of older workers. Table 6 shows our findings in levels and Table 7 shows effects in per worker terms. We find that a rise in the share of older workers leads to an increase in value added and labor costs at constant average labor productivity or costs per worker.

To set the stage for this analysis, we first confirm that, based on the unbalanced panel including all INVIND firms, an increase in employment due to 55+ workers does not have any impact on the probability of firms' exit from and entry into the sample (Table A6 of the

<sup>&</sup>lt;sup>25</sup>The Full Time Equivalent wage is obtained as the ratio between the sum of wage payments received in a given employment spell in a given year and the related days worked for full time workers; for part time workers the days worked are made equivalent to the full time ones by dividing by the full to part time hours. The denominator is thus based on contractual hours but the numerator also includes payments related to overtime hours (not separately identified in our data).

<sup>&</sup>lt;sup>26</sup>Employment variations could entail a change in the observable characteristics in the three age brackets we are considering here, resulting in a change in average wage growth that is just due to a changing composition of the pool of workers. We thus run the same regression on a wage measure that is net of composition effects, equal to the constant plus the residuals of a wage regression on gender, citizenship and the individual's exact age measured in years (Table A5). Results are not affected by this adjustment for composition.

#### Appendix).<sup>27</sup>

Based on the balance sheet variables for our balanced sample (Table 6), we focus first on two measures of total labor costs. One is reconstructed from the individual workers' administrative records (INPS) and is equal to the total gross salary paid by the firm (total FTE days worked times Average FTE wage, Column 2); the other one is coming directly from the balance sheet data (Column 3). The two measures differ because labor costs from the balance sheet include expenses related to perks and benefits paid to workers and social security contributions paid by the firm, which are not included in the INPS data. Considering both measures, we find that an increase in the share of employment of older workers increased labor costs, an effect estimated to be statistically significantly different from zero in all of our specifications. The same is true for the effect of an increase in the share of older workers on total value added.

Our different OLS estimates are again very similar, while both the IV point coefficients and the standard errors increase. By design, IV estimates reduce the amount of variation used with respect to the corresponding OLS estimates, so an increase in the variance is to be expected. Given IV and OLS estimates are not statistically different from zero, we are careful in interpreting the differences in magnitudes. It is possible that the firms most affected by the increase in the mandatory retirement age were most constrained in terms of the labor supply of older workers, and hence experienced larger increase in value added and labor costs.

Finally, all specifications indicate that a rise in the employment of 55+ workers tends to raise capital investment. Only the panel data estimates are precisely estimated, while results obtained using the 2011-12 years only confirm this pattern but the estimates are not statistically significant from zero. It is likely that the unexpected shock taking place in 2012 and due to the unanticipated pension reform did not have a strong immediate impact on firms' investment plans.

Looking across the estimates, we find that - overall - the expansion in value added associated to an increase in 55+ workers is at least as large as the rise in the associated labor costs. Indeed unit labor costs<sup>28</sup> at the firm level remain unaffected.

Table 7 analyzes the response of balance sheet variables in per-worker terms. Overall, the effects of a rise in older workers' employment we find are substantially smaller and often not statistically different from zero.

The effects on labor costs per-worker are smaller compared to the ones on total labor costs for all specifications, though point estimates are generally positive. Small increases are plausible given that the share of workers with a relatively higher pay level increases. The coefficients on value added per worker tend not to be precisely estimated, and differ across specifications. There is no indication of a negative effect, and if at all the effect estimated for the reform years is positive. Across specifications and time intervals, we find a negative (albeit imprecisely estimated) effect on capital per-worker.

Taken together, these results show that an increased presence of older workers at the firm is associated with: i) an increase in employment in the other age classes, pointing to complementarity of workers between different age groups and to the fact that mature workers might be endowed with skills that are hard to replace; ii) a null or slight negative impact on wages of older workers themselves when employed in treated firms, iii) an increase in total labor costs that is in line with- if not smaller than - the one in value added; iv) no negative impact on average labor productivity, as measured by value added per worker.

In the remaining sections, we probe the sensitivity of these results and find them to be very robust.

<sup>&</sup>lt;sup>27</sup>Additional analysis on firms' exit, for the *population* of Italian firms is reported in Table A8.

<sup>&</sup>lt;sup>28</sup>Defined as the ratio between total labor costs and value added.

# 7 Robustness

#### 7.1 Dynamic Results

In this section we estimate a set of 5 equations that are the dynamic versions of equation 3, estimated via both OLS and IV. The shock is the time-invarying-change in 55+ employment taking place in 2012 (the reform year) and the dependent variable is the cumulative change with respect to the year of the reform (2011). Since the reform was passed in December, we refer to the change with respect to 2012 as the immediate impact (the same discussed in Section 6). We refer to years 2013 and 2014 as the first and second year after the reform. The estimated coefficients on the change in the share of older workers in 2012 for these two years show the cumulative effects of the reform one or two years later. The coefficients on the models estimated for the two years before the reform (2010 and 2009) should not be statistically different from zero to confirm that, conditional on the controls, the increase in 55+ workers due to the reform should not be correlated with variations in the dependent variable that were already in place *before* the reform was introduced in 2012.

As before, we estimate the same set of regressions with OLS and IV specifications. Again, the point estimates of OLS and IV are generally qualitatively similar for different years. Not surprisingly the precision of the IV estimate declines for years other than the reform, and hence looking at both sets of estimates together provides a clearer picture of the dynamic effects. For all the main outcomes we report the corresponding graphs in Figures A5-A20 of the Appendix.

When we analyze employment outcomes across age classes, we find no different *pre trends* and precisely estimated increases in employment of younger and middle age workers that persist for up to two years after the year of the reform (Figures A5, A6 and A7 of the Appendix). The estimated effects are not statistically different in OLS and IV models. These estimates also confirm negligible effects on wages across age classes (Figures A8, A9 and A10).

As for firm outcomes, the pattern of the two measures of total labor costs found on impact is confirmed in the two years after the reform, while no differential *pre trends* are detected (Figures A11 and A12). The significant and persistent increase in total labor costs (similar in the two measures) is found to be somewhat smaller than the increase in total value added (Figure A13). We do not find a significant increase in total capital, and this is true also one or two years after the reform (Figure A14). Coherently, we find a slightly negative impact on capital per worker (Figure A15). Overall, these results confirm that the firm is able to increase production by taking advantage of the positive labor supply shock on 55+ workers without a rise in per worker labor costs or a decline worker productivity (Figures A16 to A20).

#### 7.2 Heterogeneity

We consider several dimensions of heterogeneity across firms in order to shed light on the mechanisms driving the results found and to test their robustness.

In particular, we separate *young* from *old* firms, where firms are defined as *young* if their share of 55+ workers was below the median (10%) in 2011 and *old* otherwise. Similarly we define as *small* a firm whose size was below the median (170 employees) in 2011 and large otherwise.

Results of the IV estimates<sup>29</sup> – reported in Table 8 – show that the main patterns found on the whole sample are broadly confirmed in the subgroups. Even if the standard errors become large as the number of observations shrinks in the subgroups, some interesting patterns emerge: the positive effect on youth employment is larger in magnitude (but very imprecisely estimated) in relatively younger firms. Also, the positive effect on employment seems to be concentrated

 $<sup>^{29}</sup>$ First stage statistics are reported in Table 1.

exclusively in relative larger firms, that are probably better able to adapt to an unexpected increase in the presence of 55+ workers by exploiting their internal labor market and expanding the scale of production.

#### 7.3 An alternative shock measure

According to the results of this paper, an increase in older workers' presence in the firm can be absorbed fairly well by – relatively large – Italian firms. A positive shock on the labor supply of older workers determines – if anything – a reduction in their wages, and an increase in employment levels for workers of other ages. Higher employment levels are matched by an increase in value added. These results point to the presence of complementarity between workers of different age and to the existence of specific human capital generating replacement frictions such that reductions in the retirement flows affect firms' labor demand and outcomes. Indeed, we do not find any negative impact on value added per worker of an aging pool of employees.

The absence of negative employment effects on younger workers is in line with most of the literature looking at the possible crowding out effects generated by the labor supply shocks of specific segments of the population (female participation, migration).

Using similar data and exploiting the variation in older workers employment due to the 2012 reform, but focusing on smaller firms, two papers (Boeri et al. (2017) and Bovini and Paradisi (2019), later subsumed in Bianchi et al. (2020)) do find a trade off between older and younger workers in the short run. Both papers estimate the impact of the 2012 reform in a DD setting at the firm level. In Boeri et al. (2017) the main treatment is equal to the share of 55+ workers who had to postpone retirement over all 55+ employment; in Bianchi et al. (2020), it is equal to the average number of years of minimum retirement age delay for those individuals that, at the end of 2011, were at most 3 years far from being eligible to receive a public pension (defined as Close To Retirement, CTR); the treatment is divided by the number of individuals close to retirement. In both cases, the measure of the treatment is standardized by some measure of the number of individuals that were at risk of being treated. As such, the treatment only measures the intensity of the shock on a subgroup of workers employed at the firm, but not how relevant such shock is compared to the whole workforce employed at the firm.

As a further robustness check for our main results, we carry out a last set of estimates, that take into account also the number of years of unexpected pension delay and not only the share of workers affected. The IV estimates presented in Section 5.2 rely on the fact that, due to an unexpected reform taking place in 2012, a fraction of older workers had to postpone their retirement from that year to a later time. Nevertheless, the reform could have delayed the minimum retirement age by more than a year and also for individuals who were not expected to retire in 2012 but in the near future. We thus focus also on another measure of the shock implied by the reform, that is equal to the number of years of retirement delay for those individuals that were expected to retire in the near future, divided by the 2010 number of workers employed at the firm:<sup>30</sup>

$$T_j^{2012} = \frac{\sum_i YTR_{2012,i,j} | Law_{2012} - \sum_i YTR_{2012,i,j} | Law_{2011}}{EMPL_{2011,j}}$$
(6)

The variable YTR identifies Years To Retirement for workers i that, according to the law in place just before the 2012 reform were alternatively within one and five years from their MRA.

<sup>&</sup>lt;sup>30</sup>This specification is close to the one adopted by Bovini and Paradisi (2019) later subsumed in Bianchi et al. (2020), with the exception that the shock is equal to the total number of years of pension eligibility delay among Close To Retirement (CTR) individuals divided by total employment rather than by the number of CTRs.

We thus estimate the following equation for year 2012-11:

$$\Delta y_{j,t} = \alpha + d_{st} + \beta * T_j^{2012} + \phi X_{j,2010} + \epsilon_{jt} \tag{7}$$

where  $T_j^{2012}$  is the treatment as defined in equation 6,  $\Delta y_{j,t}$  is the year on year change in the dependent variable,  $d_s$  are sector fixed effects, and  $X_{j,2010}$  are the usual controls (shares of 35-54 and 55+ workers in 2010).

First of all, we look at the impact of the treatment  $T_j^{2012}$  on employment variation of 55+ workers themselves, finding a positive coefficient irrespective of the number of YTR used to define the group of close to retirement individuals (Table 9). Positive effects of the treatment on employment are confirmed also for the other age classes (15-34 and 35-54). Results are confirmed as well when looking at Full Time Equivalent wages, and for all of the firm outcomes.

#### 7.4 Additional evidence on the population of Italian firms

As a final robustness check we estimate equation 3 on an alternative dataset having partial info on the *population* of Italian firms (we are thus working with 1.5 million firms for the 2012-11 period). We do not use this dataset in the rest of the analysis because it does not include information on the years of social security contributions each worker has paid and – as such – it does not provide the information necessary to simulate the actual and expected MRA. For these reasons, the additional estimates cannot rely on an IV strategy. We estimate the first differences OLS specification of equation 3 in the reform years (2011-2012).

Results reported in Table A8 are again in line with those obtained with our sample of relatively large firms and show a positive correlation between 55+ employment variation and employment variation in other age classes, a negligible impact on wages and positive effects on capital and value added.

## 8 Conclusions

In response to population aging, in the last decades many governments provided incentives to postpone retirement through increases in the statutory retirement age or through tax breaks. However, the increased participation of older workers has raised two main concerns. There are fears that older workers may crowd out younger cohorts in the labor market, by reducing their work opportunities. A larger presence of older workers may hamper firms' productivity and future growth since older workers, even if more experienced, may be less innovative and less willing to take risks than younger cohorts.

This paper analyzes the causal effects of a short-run increase in employment of older workers on firms' input mix, wages, labor costs, total capital, value added, and average labor productivity. After showing that in the Italian institutional setting older workers mostly retire after reaching pension age, we exploit a recent reform of pension eligibility as an exogenous increase in the presence of older workers at the firm level. Effects are estimated based on a unique matched employer, employee, balance sheet data set for the period 2010-2014.

We find that an unexpected increase in the share of older workers leads to a positive impact on young and middle-age employment. An exogenous 10% increase in the number of old workers implies a 1.8% increase in the number of young and 1.3% of middle-aged workers. Total labor costs increase in line with employment, and remain broadly constant in per-worker terms. Total value added increases and per worker labor productivity is constant. These adjustments occur with little changes in daily wages. Despite substantial aggregate employment responses, we also find no evidence of aggregate wages for older workers. These results are consistent with a model in which firm-specific human capital or market frictions make it costly to replace older, higher-tenured workers in the market. Younger and middle-aged workers appear to be imperfect substitutes, and in our setting are likely complements to older workers. The fact that labor costs increase in line with employment and tend to respond less than value added signals that firms were able to adjust smoothly, and may have benefited, from the shock implied by the pension reform. There is no evidence that older workers may have been overpaid relative to their productivity, or that their increase implied a burden to the firm. Overall, the absence of any negative impact on value added per worker seems to point out that concerns on the impact of an aging workforce on productivity might be overstated, in line with the findings of Acemoglu and Restrepo (2018).

Generalizing our results requires some caution. First, for data reasons, in our main results we focus only on large firms, which are less likely to be credit constrained and perhaps more able to expand employment. While we show our findings are robust for a broader sample that includes smaller firms, a separate analysis with high-quality economic data for smaller firms may be fruitful for future work. Second, since the effective retirement age in Italy is not very high (62 years old for men and 61 for women; corresponding figures at OECD level are 65 and 63.6, respectively), the non-negative effects on firms' outcomes we found might be driven by the fact that incumbent older workers are not too old and thus still productive. Finally, by design, our analysis is neither equipped to capture market-level responses or to isolate longer-term relationships between population aging and economic outcomes.

# References

Acemoglu, D. and P. Restrepo (2018). Demographics and automation. *mimeo*.

- Adamopoulou, E., E. Bobbio, M. D. Philippis, and F. Giorgi (2016). Wage rigidities and business cycle fluctuations: a linked employer-employee analysis. *IZA Journal of Labor Policy* 22(5), 164–194.
- Attanasio, O., S. Kitao, and G. L. Violante (2007). Global demographic trends and social security reform. *Journal of Monetary Economics* 54(1), 144–198.
- Azoulay, P., J. Wang, and J. Gra Zivin (2010). Superstar extinction. The Quarterly Journal of Economics 125(2), 549–589.
- Bank of Italy (2014). Survey of Industrial and Service Firms. Technical report, Bank of Italy.
- Battistin, E., A. Brugiavini, E. Rettore, and G. Weber (2009). The retirement consumption puzzle: evidence from a regression discontinuity approach. *The American Economic Review* 99(5), 2209–2226.
- Becker, G. S. (1962). Investment in human capital: A theoretical analysis. Journal of Political Economy 70(5, Part 2), 9–49.
- Becker, G. S. (1964). *Human capital: a theoretical analysis with special reference to education*. New York: Columbia University Press.
- Ben-Porath, Y. (1967). The production of human capital and the life cycle of earnings. *Journal* of Political Economy 75(4, Part 1), 352–365.
- Bertoni, M. and G. Brunello. Does Delayed Retirement Affect Youth Employment? Evidence from Italian Local Labour Markets. *Economic Policy*.
- Bianchi, N., G. Bovini, J. Li, M. Paradisi, and M. Powell (2020). Career spillovers in internal labor markets. *mimeo*.
- Boeri, T., P. Garibaldi, and E. Moen (2017). Closing the Retirement Door and the Lump of Labor. *mimeo*.
- Bovini, G. and M. Paradisi (2019). Labor substitutability and the impact of raising the retirement age. *WorkINPS paper*.
- Bowlus, A. J. and C. Robinson (2012). Human capital prices, productivity, and growth. *American Economic Review* 102(7), 3483–3515.
- Cardoso, A. R., P. Guimarães, and J. Varejão (2011). Are older workers worthy of their pay? An empirical investigation of age-productivity and age-wage nexuses. *De Economist* 159(2), 95.
- Cataldi, A., S. Kampelmann, and F. Rycx (2012). Does it pay to be productive? The case of age groups. *International Journal of Manpower* 33(3), 264–283.
- Covip (2018). Relazione annuale per l'anno 2017. Technical report, Commissione di Vigilanza sui Fondi Pensione.
- Cribb, J., C. Emmerson, and G. Tetlow (2016). Signals matter? Large retirement responses to limited financial incentives. *Labour Economics* (42), 203–212.

- D'Aurizio, L. and G. Papadia (2019). Using administrative data to evaluate sampling bias in a business panel survey. *Journal of Official Statistics* 35(1), 67–92.
- Daveri, F. and M. Maliranta (2007). Age, seniority and labour costs: lessons from the Finnish IT revolution. *Economic Policy* 22(49), 118–175.
- Doran, K., A. Gelber, and A. Isen (2015). The effects of high-skilled immigration policy on firms: Evidence from visa lotteries. *NBER WP* (20668).
- Dostie, B. (2011). Wages, productivity and aging. De Economist 159(2), 139–158.
- Engbom, N. (2019). Firm and worker dynamics in an aging labor market. *FED Minneapolis* WP (756).
- Feyrer, J. (2007). Demographics and productivity. The Review of Economics and Statistics 89(1), 100–109.
- Gathmann, C. and U. Schönberg (2010). How general is human capital? A task-based approach. Journal of Labor Economics 28(1), 1–49.
- Göbel, C. and T. Zwick (2012). Age and productivity: sector differences. *De Economist 160*(1), 35–57.
- Gruber, J. and D. A. Wise (2010). Social security programs and retirement around the world: The relationship to youth employment. University of Chicago Press.
- Haltiwanger, J. C., J. I. Lane, and J. R. Spletzer (1999). Productivity differences across employers: The roles of employer size, age, and human capital. *The American Economic Review* 89(2), 94–98.
- Hansen, A. H. (1939). Economic progress and declining population growth. The American Economic Review 29(1), 1–15.
- Hut, S. (2019). Financial frictions and firm labor adjustments: Evidence from a retirement policy. *mimeo*.
- Ilmakunnas, P. and S. Ilmakunnas (2015). Hiring older employees: do the age limits of early retirement and the contribution rates of firms matter? The Scandinavian Journal of Economics 117(1), 164–194.
- Isen, A. (2013). Dying to know: Are workers paid their marginal product. University of Pennsylvania Working Paper.
- Italian Institute of Social Security (2016). XV Annual Report. Technical report.
- Jaeger, S. and J. Heining (2020). How Substitutable Are Workers? Evidence from Worker Deaths. *MIT*, *mimeo*.
- Jaravel, X., N. Petkova, and A. Bell (2018). Team-specific capital and innovation. American Economic Review 108(4-5), 1034–73.
- Lalive, R., A. Magesan, and S. Staubli (2017). Raising the full retirement age: Defaults vs incentives.
- Lallemand, T. and F. Rycx (2009). Are older workers harmful for firm productivity? *De Economist* 157(3), 273–292.

- Lazear, E. P. (2009). Firm-specific human capital: A skill-weights approach. Journal of Political Economy 117(5), 914–940.
- Maestas, N., K. J. Mullen, and D. Powell (2016). The effect of population aging on economic growth, the labor force and productivity. (22452).
- Mahlberg, B., I. Freund, J. C. Cuaresma, and A. Prskawetz (2013). Ageing, productivity and wages in Austria. *Labour Economics* 22, 5–15.
- Mahlberg, B., I. Freund, and A. Prskawetz (2009). Firm productivity, workforce age and vocational training in Austria. *Labour Markets and Demographic Change*, 58–84.
- Malmberg, B., T. Lindh, and M. Halvarsson (2008). Productivity consequences of workforce aging: stagnation or Horndal effect? *Population and Development Review* 34, 238–256.
- Manoli, D. S. and A. Weber (2016). The effects of the early retirement age on retirement decisions. *NBER WP* (22561).
- Martins, P. S., A. A. Novo, and P. Portugal (2009). Increasing the legal retirement age: The impact on wages, worker flows and firm performance.
- Mastrobuoni, G. (2009). Labor supply effects of the recent social security benefit cuts: Empirical estimates using cohort discontinuities. *Journal of Public Economics* 93(11), 1224–1233.
- Mohnen, P. (2019). The Impact of the Retirement Slowdown on the U.S. Youth Labor Market. *mimeo*.
- Moody's Investor Service (2010). Population aging will dampen economic growth over the next two decades. Technical report, February 4, 2010.
- Neal, D. (1995). Industry-specific human capital: Evidence from displaced workers. Journal of Labor Economics 13(4), 653–677.
- OECD (2015). Ageing and Employment Policies Statistics on average effective age of retirement.
- Parent, D. (2000). Industry-specific capital and the wage profile: Evidence from the national longitudinal survey of youth and the panel study of income dynamics. *Journal of Labor Economics* 18(2), 306–323.
- Poletaev, M. and C. Robinson (2008). Human capital specificity: evidence from the dictionary of occupational titles and displaced worker surveys, 1984–2000. Journal of Labor Economics 26(3), 387–420.
- Sauvagnat, J. and F. Schivardi. Are Executives in Short Supply? Evidence from Deaths' Events. *mimeo*.
- Sestito, P. and E. Viviano (2018). Firing costs and firm hiring: evidence from an Italian reform. *Economic Policy* 33(93), 101–130.
- Staubli, S. and J. Zweimüller (2013). Does raising the early retirement age increase employment of older workers? *Journal of Public Economics 108*, 17–32.
- Storesletten, K. (2000). Sustaining fiscal policy through immigration. Journal of Political Economy 108(2), 300–323.

- Summers, L. (2013). Why stagnation might prove to be the new normal. Technical report, The Financial Times.
- Tommasino, P. and R. Zizza (2015). The lump of labour fallacy: a reassessment for the euro area. *Occasional Paper Series 159*.
- Topel, R. (1990). Specific capital and unemployment: Measuring the costs and consequences of job loss. In *Carnegie-Rochester conference series on public policy*, Volume 33, pp. 181–214. Elsevier.
- Van Ours, J. C. and L. Stoeldraijer (2011). Age, wage and productivity in dutch manufacturing. De Economist 159(2), 113–137.
- Vandenberghe, V. (2013). Are firms willing to employ a greying and feminizing workforce? Labour Economics 22, 30–46.
- Waldinger, F. (2012). Peer Effects in Science: Evidence from the Dismissal of Scientists in Nazi Germany. *The Review of Economic Studies* 79(2), 838–861.

Figures and Tables



Figure 1: Public pension eligibility rules, by type

Notes: YSSC stands for Years of Social Security Contributions. Under the seniority scheme, before the 2011 reform the eligibility rules were the same for men and women. One option referred only to the number of accrued YSSC (*YSSC pre-2012*); the second one referred to a combination of requirements on the number of YSSC and age (*YSSC Quota pre-2012*). After the reform, the second option - the Quota - was abolished. The seniority pension scheme requires a given number of accrued YSSC which differs across genders.



Figure 2: Exptected and actual public pension eligibility and MRA

Notes: Our calculations based on INPS-INVIND data. Solid line: share of workers who are eligible for a public pension in year t according to the law in place in year t,  $ELIG_{i,t}|Law_t$ . Dotted line: expected share of workers who are eligible for a public pension in year t according to the law in place in year t - 1,  $ELIG_{i,t}|Law_{t-1}$ . The solid vertical line identifies the first year in which the pension reform we analyze was implemented.



Notes: Our calculations based on INPS-INVIND data. Solid line: Average Minimum Retirement Age in year t according to the law in place in year t,  $MRA_{i,t}|Law_t$ . Dotted line: expected share of workers who are eligible for a public pension in year t according to the law in place in year t - 1,  $MRA_{i,t}|Law_{t-1}$ . The solid vertical line marks the first year in which the pension reform we analyze was implemented.

Figure 3: The evolution of the share of 55+ workers (actual and pre-2012 trend)



Notes: Our calculations on INVIND sample used in the analysis, see Table ?? for descriptive statistics. In dark grey: the actual share of 55+ workers; in light grey: the predicted line over the pre-2012 trend.



Figure 4: Claiming and retirement around pension eligibility

Conditional on gender age and YSSC

Notes: Our calculations on INPS data. Results of an auxiliary regression estimating the probability to claim pension benefits as a function of the time distance (measured in years) to retirement eligibility. Point estimates and 95% confidence intervals. Information on actual benefit claim is available only for a subsample of workers.



Conditional on gender age and YSSC

Notes: Our calculations on INPS data. Results of an auxiliary regression estimating the probability of being employed as function of the time distance (measured in years) to retirement eligibility. Point estimates and 95% confidence intervals. Information on actual benefit claim is available only for a subsample of workers.

Figure 5: 55+ workers unexpectedly losing eligibility in 2012: firm level share on total workforce



 $\it Notes:$  Our calculations on INPS data.

Figure 6: The 2012 shock on 55+ employment: pre-trends and dynamic effects



*Notes*: The graph reports estimates of a regression of the instrumented 55+ employment change in 2012 as in equation 5 of Subsection 5.2.1 on up to two leads and two lags of cumulative 55+ employment change.



Figure 7: The wage-age gradient for mature workers in the overall economy

Notes: Our calculations on INPS data. Data cover the whole private non agricultural sector; only employees working at least 150 days in a given year are included. Each wage level is normalized to the average for 40-44 years old in each year.

2011 - 2012
Years
regression.
stage
$\operatorname{First}$
÷
Table

	All fir	sm	Your	1g	Olc		Sma	II	Bi	50
	Coef.	S.e.	Coef.	S.e.	Coef.	S.e.	Coef.	S.e.	Coef.	S.e.
$\frac{ELIG_{j,2012} Law_{2011}}{EMPL_{j,2011}}$	-0.39***	(0.10)	-0.38***	(0.08)	-0.40***	(0.12)	-0.35***	(0.11)	-0.53***	(0.22)
$\frac{ELIG_{j,2012}  \tilde{L}aw_{2011} - ELIG_{j,2012}  Law_{2012}}{EMPL_{i,2011}}$	$0.56^{***}$	(0.18)	$0.32^{***}$	(0.12)	$0.61^{***}$	(0.22)	$0.56^{***}$	(0.22)	$0.69^{***}$	(0.38)
Obs	102	5 L	437		588		202		518	~
F-test	31.9	6	9.72	~1	19.2	4	16.7	~	21.4	<u>5</u>
Notes: First stage r	egression (e	quation	<b>5</b> ). Depend	ent varia	ble:					

o). Depe n ho \ 50 λ D

# $\frac{OLD_{j,2012} - OLD_{j,2011}}{EMPL_{j,2011}}$

The average predicted change in employment due to the increase 55+ workers caused by the reform

is equal to 0.4 percentage points. \* significant at 10%; \*\* significant at 5%; \*\*\* significant at 1%.

	15-34	35-54	15-34	35-54
	Balance	ed panel	Unbalan	ced panel
FD OLS				
Net empl var $55+$	0.391***	$1.466^{***}$	$0.463^{***}$	$1.694^{***}$
	(0.0419)	(0.140)	(0.0493)	(0.149)
Obs	5125	5125	7101	7101
FD OLS with FE				
Net empl var $55+$	0.364***	$1.467^{***}$	$0.411^{***}$	$1.610^{***}$
	(0.0436)	(0.160)	(0.0451)	(0.154)
Obs	5125	5125	7101	7101
FD OLS (Years 2011-12)				
	0.407***	$1.629^{***}$	$0.557^{***}$	$1.700^{***}$
	(0.0547)	(0.268)	(0.0735)	(0.184)
Obs	1025	1025	1960	1960
FD IV (Years 2011-12)				
Net empl var 55+	0.522**	$0.786^{*}$	0.557	$0.848^{**}$
	(0.262)	(0.405)	(0.344)	(0.401)
Obs	1025	1025	1960	1960

Table 2: Effects on employment, by age class. Years: 2010-2014 or 2012-11

*Notes:* Estimates of equation 3 on the balanced panel. Regressions include as additional controls: the share of 35-54 and 55+ workers in 2010, sector fixed effects and year fixed effects (not included in the one year regressions 2011-12). First stage statistics for the IV estimates are reported in Table 1. Standard errors in brackets clustered at the firm level. \* significant at 10%; \*\* significant at 5%; \*\*\* significant at 1%.

		Hiring rate		Se	paration rat	e
	15-34	35 - 54	55 +	15-34	35-54	55 +
FD OLS						
Net empl var $55+$	0.441***	$1.197^{***}$	$0.352^{***}$	0.0501	-0.269***	$-0.648^{***}$
	(0.0553)	(0.131)	(0.0439)	(0.0443)	(0.0978)	(0.0439)
Obs	5125	5125	5125	5125	5125	5125
FD OLS with FE						
Net empl var $55+$	0.280***	$0.895^{***}$	$0.281^{***}$	-0.0847***	-0.572***	-0.719***
	(0.0414)	(0.133)	(0.0469)	(0.0287)	(0.0808)	(0.0469)
Obs	5125	5125	5125	5125	5125	5125
FD OLS (Years 2011-12)						
	0.409***	$1.541^{***}$	$0.536^{***}$	0.00152	-0.0883	-0.464***
	(0.0802)	(0.295)	(0.136)	(0.0707)	(0.124)	(0.136)
Obs	1025	1025	1025	1025	1025	1025
FD IV (Years 2011-12)						
Net empl var $55+$	0.480	0.484	0.102	-0.0422	-0.302	-0.898***
	(0.498)	(0.649)	(0.256)	(0.437)	(0.565)	(0.256)
Obs	1025	1025	1025	1025	1025	1025

Table 3: Effects hirings and separations, by age. Years: 2010-2014.

*Notes:* Estimates of equation 3. Regressions include as additional controls: the share of 35-54 and 55+ workers in 2010, sector fixed effects and year fixed effects (not included in the one year regressions 2011-12). First stage statistics for the IV estimates are reported in Table 1. Standard errors in brackets clustered at the firm level.

	Fixe	ed term con	tract	Perr	nanent cont	ract
	15-34	35 - 54	55 +	15-34	35 - 54	55 +
FD OLS	$0.125^{***}$	$0.167^{***}$	$0.0866^{***}$	$0.266^{***}$	$1.300^{***}$	$0.913^{***}$
Net empl var $55+$	(0.0218)	(0.0300)	(0.0173)	(0.0352)	(0.138)	(0.0173)
Obs	5125	5125	5125	5125	5125	5125
FD OLS with FE						
Net empl var $55+$	$0.107^{***}$	$0.154^{***}$	$0.0950^{***}$	$0.257^{***}$	$1.313^{***}$	$0.905^{***}$
	(0.0200)	(0.0331)	(0.0200)	(0.0402)	(0.158)	(0.0200)
Obs.	5125	5125	5125	5125	5125	5125
FD OLS (Years 2011-12)						
	$0.112^{***}$	$0.187^{***}$	$0.0552^{***}$	$0.295^{***}$	$1.442^{***}$	$0.945^{***}$
	(0.0352)	(0.0395)	(0.0145)	(0.0561)	(0.260)	(0.0145)
Obs	1025	1025	1025	1025	1025	1025
FD IV (Years 2011-12)						
Net empl var $55+$	$0.332^{*}$	-0.120	0.0483	0.190	$0.906^{**}$	$0.952^{***}$
	(0.188)	(0.181)	(0.0545)	(0.167)	(0.359)	(0.0545)
	1025	1025	1025	1025	1025	1025

#### Table 4: Net employment variation by contract type. Years: 2010-2014.

*Notes:* Estimates of equation 3. Regressions include as additional controls: the share of 35-54 and 55+ workers in 2010, sector fixed effects and year fixed effects (not included in the one year regressions 2011-12). First stage statistics for the IV estimates are reported in Table 1. Standard errors in brackets clustered at the firm level.

\* significant at 10%; \*\* significant at 5%; \*\*\* significant at 1%.

Table 5:	Effects	on FTE	wages,	by	age class.	Years:	2010-2014.
----------	---------	--------	--------	----	------------	--------	------------

VARIABLES	15-34	35 - 54	55 +
FD OLS	-0.057	-0.036	-0.507***
Net empl var $55+$	(0.0567)	(0.045)	(0.105)
	5113	5125	5088
Obs.			
FD OLS with firm FE			
Net empl var $55+$	-0.044	-0.030	-0.506***
	(0.065)	(0.046)	(0.111)
Obs.	5113	5125	5088
FD (Years 2011-12)			
	-0.100	-0.008	-0.363
	(0.168)	(0.104)	(0.243)
	1022	1025	1020
FD IV (Years 2011-12)			
Net empl var $55+$	-0.132	0.0926	-0.456
	(0.596)	(0.433)	(1.005)
Obs.	1022	1025	1020

*Notes:* Estimates of equation 3. Regressions include as additional controls: the share of 35-54 and 55+ workers in 2010, sector fixed effects and year fixed effects (not included in the one year regressions 2011-12). First stage statistics for the IV estimates are reported in Table 1. Standard errors in brackets clustered at the firm level.

					$\mathbf{Unit}$	
	La	bor cost	Value	La	bor costs	
VARIABLES	INPS	Balance sheet	Added	INPS	Balance sheet	Capital
			1			1
FD OLS	0.773***	0.682***	0.729***	0.0255	-0.0448	0.434***
Net empl var 55+	(0.0594)	(0.0762)	(0.115)	(0.0968)	(0.0824)	(0.104)
	5116	5116	5116	5116	5116	5116
Obs						
FD OLS with FE						
Net empl var 55+	$0.499^{***}$	$0.403^{***}$	0.415***	0.0841	-0.00623	0.283***
	(0.0600)	(0.0757)	(0.136)	(0.116)	(0.102)	(0.0999)
Obs	5116	5116	5116	5116	5116	5116
FD OLS (Years 2011-12)						
	$0.645^{***}$	$0.595^{***}$	0.929***	-0.148	-0.258*	0.197
	(0.0857)	(0.104)	(0.190)	(0.169)	(0.156)	(0.195)
Obs	1025	1025	1025	1025	1025	1025
FD IV (Years 2011-12)	$1.678^{***}$	$1.585^{**}$	$3.767^{**}$	-0.217	-0.481	0.706
Net empl var 55+	(0.609)	(0.640)	(1.759)	(1.139)	(1.035)	(0.982)
Obs	1025	1025	1025	1025	1025	1025

#### Table 6: Firm outcomes. Years: 2010-2014.

*Notes*: Estimates of equation 3. Labor cost-INPS is the total labor cost obtained by multiplying the average wage by the number of FTE working days in INPS data. Labor cost-Balance sheet is the total labor cost according to the firm balance sheet. Regressions include as additional controls: the share of 35-54 and 55+ workers in 2010, sector fixed effects and year fixed effects (not included in the one year regressions 2011-12). First stage statistics for the IV estimates are reported in Table 1. Standard errors in brackets clustered at the firm level.

\* significant at 10%; \*\* significant at 5%; \*\*\* significant at 1%.

VARIABLES	Capital	La INPS	bor cost Balance sheet	Value Added
FD OLS	-0.291**	0.109***	0.0720	0.0572
Net empl var 55+	(0.127)	(0.0368)	(0.0533)	(0.106)
Obs	5116	5116	5116	5116
<b>FD OLS with FE</b> Net empl var 55+ Obs	-0.159 (0.137) 5116	$0.0938^{**}$ (0.0443) 5116	$\begin{array}{c} 0.0474 \\ (0.0629) \\ 5116 \end{array}$	$\begin{array}{c} -0.0389 \\ (0.129) \\ 5116 \end{array}$
FD OLS (Years 2011-12)	-0.143	0.228***	0.225**	0.461**
Obs	$(0.219) \\ 1025$	(0.0637) 1025	(0.0901) 1025	(0.190) 1025
FD IV (Years 2011-12)	-0.473	0.327	0.213	1.610
Net empl var 55+	(1.305)	(0.379)	(0.507)	(1.396)
Obs	1025	1025	1025	1025

Table 7: Firm outcomes - per worker levels. Years 2010-14

*Notes:* Estimates of equation 3. Labor cost-INPS is the total labor cost obtained by multiplying the average wage by the number of FTE working days in INPS data. Labor cost-Balance sheet is the total labor cost according to the firm balance sheet. Regressions include as additional controls: the share of 35-54 and 55+ workers in 2010, sector fixed effects and year fixed effects (not included in the one year regressions 2011-12). Standard errors in brackets clustered at the firm level. First stage statistics for the IV estimates are reported in Table 1.

Heterogeneity
$\ddot{\infty}$
Table

	Net en	npl var		Wage var		Labor	: cost	Value	added
VARIABLES	15-34	35-54	15 - 34	35-54	55+	$\operatorname{Tot}$	p.w.	$\operatorname{Tot}$	p.w.
YOUNG									
Net empl var $55+$	1.367	0.350	-2.157	1.748	-1.037	$5.382^{*}$	$3.940^{*}$	$8.852^{*}$	6.420
	(1.247)	(1.306)	(2.235)	(1.929)	(5.896)	(3.120)	(2.336)	(5.371)	(5.002)
Observations OLD	437	437	437	437	432	437	437	437	437
Net empl var 55+	$0.441^{*}$	$0.840^{*}$	0.136	-0.213	-0.535	$1.126^{**}$	-0.181	$3.025^{*}$	0.944
	(0.237)	(0.431)	(0.610)	(0.454)	(0.877)	(0.570)	(0.516)	(1.721)	(1.355)
Observations	588	588	585	588	588	588	588	588	588
SMALL									
Net empl var $55+$	0.299	0.438	-0.297	0.204	-0.234	$1.574^{**}$	0.201	2.743	0.647
	(0.285)	(0.555)	(0.735)	(0.517)	(1.209)	(0.729)	(0.599)	(1.774)	(1.496)
Observations	507	507	504	507	503	507	507	507	507
LARGE									
Net $empl var 55+$	$1.090^{*}$	$1.828^{**}$	0.260	-0.0507	-2.154	1.062	-0.129	6.141	4.121
	(0.627)	(0.931)	(0.815)	(0.654)	(1.512)	(0.945)	(0.770)	(3.909)	(2.978)
Observations	518	518	518	518	517	518	518	518	518
Notes: Estimates	of equation	3. Regressi	ons include	e as addition	al controls:	the share e	of 35-54 and	155+ work	ers in
$2010$ , sector fixed $\epsilon$	effects. All	dependent v	ariables co	me from bal	ance sheet	data (CEBI	or INVIND	). All regre	ssions
-	-	- - -	1 1 1	1	0100		-	, - ,	

include sector and year fixed effects, the share of 35-54 and 55+ in 2010. Standard errors in brackets clustered at the firm level. \* significant at 10%; \*\* significant at 5%; \*\*\* significant at 1%.

		V = VT	R <= 1	T = Y = 0	$R \leq = 2$	$0 \ll Y = YT$	R <= 3	$0 \ll YT$	R <= 4	$0 \leq = YT$	R <= 5
Net empl var	15-34	0.058	(0.047)	0.0423	(0.028)	$0.036^{*}$	(0.020)	$0.029^{*}$	(0.015)	$0.038^{***}$	(0.011)
	35-54	$0.327^{**}$	(0.127)	$0.172^{*}$	(0.093)	$0.086^{**}$	(0.037)	$0.078^{**}$	(0.033)	$0.079^{***}$	(0.026)
	55+	$0.164^{***}$	(0.058)	$0.100^{**}$	(0.043)	$0.069^{***}$	(0.018)	$0.059^{***}$	(0.015)	$0.044^{***}$	(0.011)
Delta wage	15-34	0.054	(0.086)	0.041	(0.051)	-0.013	(0.031)	0.011	(0.026)	0.010	(0.019)
	35-54	0.005	(0.063)	0.046	(0.034)	0.015	(0.022)	0.024	(0.018)	0.018	(0.015)
	55+	0.162	(0.163)	-0.027	(0.078)	-0.017	(0.057)	0.004	(0.046)	-0.022	(0.039)
Firm outcomes	Total										
Capital		$0.392^{**}$	(0.167)	0.123	(0.079)	$0.117^{**}$	(0.056)	0.071	(0.048)	$0.070^{*}$	(0.039)
Labor cost	INPS	$0.287^{**}$	(0.115)	$0.269^{***}$	(0.059)	$0.168^{***}$	(0.035)	$0.155^{***}$	(0.031)	$0.154^{***}$	(0.029)
Labor cost	Balance sheet	0.257**	(0.125)	$0.252^{***}$	(0.057)	$0.177^{***}$	(0.036)	$0.149^{***}$	(0.0312)	$0.148^{***}$	(0.0268)
Value added		0.301	(0.245)	$0.320^{***}$	(0.121)	$0.254^{***}$	(0.084)	$0.246^{***}$	(0.068)	$0.244^{***}$	(0.055)
Firm outcomes	Per worker										
Capital		0.367	(0.232)	-0.080	(0.112)	-0.031	(0.081)	-0.069	(0.069)	-0.051	(0.058)
Labor cost	INPS	$0.137^{**}$	(0.0665)	$0.0875^{**}$	(0.0346)	0.0267	(0.0225)	0.0218	(0.0189)	0.023	(0.016)
Labor cost	Balance sheet	0.120	(0.088)	0.061	(0.043)	0.013	(0.031)	0.0022	(0.026)	0.009	(0.020)
Value added		0.158	(0.193)	0.086	(0.110)	0.051	(0.078)	0.0682	(0.063)	$0.088^{*}$	(0.049)
Estimates of equation 7	. Regressions includ	e as additio	nal controls	: the share	of 35-54 an	d 55+ work	ers in 2010,	sector fixed	effects. Al	l regressions	include sector and

measure	
$\operatorname{shock}$	
alternative	
$\mathbf{An}$	
Table 9:	

n ⊃ *Notes:* Estimates of equation 7. Regressions include as additional controls: the share of 35-54 and 55+ work year fixed effects, the share of 35-54 and 55+ in 2010. Standard errors in brackets clustered at the firm level. \* significant at 10%; \*\*\* significant at 1%.

# A Appendix

#### A.1 Details on the institutional setting

The Italian welfare system features two main public pension schemes: *old age* and *seniority*.

As for the *old age* pension scheme, before the reform the retirement age was 60 for women and 65 for men, requiring also a minimum number of accrued years of contribution.<sup>31</sup> The Fornero pension reform swiftly increased the retirement age for all workers up to 67 by 2020, both for men and for women, with at least 20 years of paid social security contributions; moreover, the reform allowed all individuals to retire at 70, as long as they accrued at least 5 years of contribution.

As for the *seniority* pension scheme, before the reform, eligibility required either 40 years of paid contributions (irrespective of age) or a mix of age and years of social security contributions, the so called "quota system" (for instance, the sum of age and years of social security contribution should have been 96 in 2011, with at least 60 years of age and 35 years of social security contribution). The Fornero reform abolished the "quota system" and raised the minimum years of paid contribution in 2012 from 40 to 42 for men, to 41 for women.<sup>32</sup> Finally, the new rules in place since 2012 allowed workers who were already eligible for a public pension when the bill passed to retire under the pre-reform rules, without losing their eligibility. This option was not available before the reform, since workers could retire in a given year only if eligible under the rules in place that given year.

Finally, early retirement is available only for women and has not been affected by the reform. Such an option, despite the increase in the take-up after the reform, is rarely used, given that it implies, on average, a 35% cut in the monthly transfer (Italian Institute of Social Security, 2016). The maximum value of the take-up rate was not more than 20% in 2015 (around 11% before the reform). There is no mandatory retirement and working after retirement is not prohibited.

 $<sup>^{31}</sup>$ Eligibility for old age scheme also required 20 accrued years of contribution. Before the Fornero reform, the requirement was of 5 years for workers who had started to work since January 1996 (under the defined contribution scheme), while it was already 20 for those who had started to work before January 1996 (under the defined benefit scheme).

 $<sup>^{32}</sup>$ In 2013, the minimum number of required years of contributions rose to 43 for men and 42 for women; from 2014 onward to 44 for men and 43 for women, respectively.

# A.2 Additional figures and tables

#### Figure A1: Google searches for "Riforma pensioni"



Notes: Downloaded from https://trends.google.it/trends/ on February, 19, 2020.



Figure A2: First stage

Notes: For each quintile of the distribution of eligibility change for 55+ workers expressed as a share of total employment, the X axis reports the average eligibility change for 55+ workers expressed as a share of total employment and the Y axis reports the average change in 55+ employment as a share of total employment.



Figure A3: A closer look at the residuals

*Notes*: The figure is based on equation 3. X axis reports - for each quintile - the average of the residuals of a regression of the net employment variation of 55+workers on controls. Y axis reports the corresponding averages for 15-34 residuals.



*Notes*: The figure is based on the IV estimates for equation 3. X axis reports - for each quintile - the average of the residuals of a regression of the *instrumented* net employment variation of 55+ workers on controls. Y axis reports the corresponding averages for 15-34 residuals.

Figure A4: Retirement flows by exact age, 55+ workers



 $\it Notes:$  Our calculations on INPS data



Figure A5: Estimated cumulative effects

Notes: Estimates of equation 3 on INPS data, see Section 7.1 for details. First stage statistics for the IV estimates are reported in Table 1. Point estimates and 95% confidence intervals.



Figure A6: Estimated cumulative effects

Notes: Estimates of equation 3 on INPS data, see Section 7.1 for details. First stage statistics for the IV estimates are reported in Table 1. Point estimates and 95% confidence intervals.



Figure A7: Estimated cumulative effects

Notes: Estimates of equation 3 on INPS data, see Section 7.1 for details. First stage statistics for the IV estimates are reported in Table 1. Point estimates and 95% confidence intervals.



Figure A8: Estimated cumulative effects

Notes: Estimates of equation 3 on INPS data, see Section 7.1 for details. First stage statistics for the IV estimates are reported in Table 1. Point estimates and 95% confidence intervals.



Figure A9: Estimated cumulative effects

Notes: Estimates of equation 3 on INPS data, see Section 7.1 for details. First stage statistics for the IV estimates are reported in Table 1. Point estimates and 95% confidence intervals.



Figure A10: Estimated cumulative effects

Notes: Estimates of equation 3 on INPS data, see Section 7.1 for details. First stage statistics for the IV estimates are reported in Table 1. Point estimates and 95% confidence intervals.



Figure A11: Estimated cumulative effects

Notes: Estimates of equation 3 on INPS data, see Section 7.1 for details. First stage statistics for the IV estimates are reported in Table 1. Point estimates and 95% confidence intervals.



Figure A12: Estimated cumulative effects

Notes: Estimates of equation 3 on INPS data, see Section 7.1 for details. First stage statistics for the IV estimates are reported in Table 1. Point estimates and 95% confidence intervals.



Figure A13: Estimated cumulative effects

Notes: Estimates of equation 3 on INPS data, see Section 7.1 for details. First stage statistics for the IV estimates are reported in Table 1. Point estimates and 95% confidence intervals.



Figure A14: Estimated cumulative effects

Notes: Estimates of equation 3 on INPS data, see Section 7.1 for details. First stage statistics for the IV estimates are reported in Table 1. Point estimates and 95% confidence intervals.



Figure A15: Estimated cumulative effects

Notes: Estimates of equation 3 on INPS data, see Section 7.1 for details. First stage statistics for the IV estimates are reported in Table 1. Point estimates and 95% confidence intervals.



Figure A16: Estimated cumulative effects

Notes: Estimates of equation 3 on INPS data, see Section 7.1 for details. First stage statistics for the IV estimates are reported in Table 1. Point estimates and 95% confidence intervals.



Figure A17: Estimated cumulative effects

Notes: Estimates of equation 3 on INPS data, see Section 7.1 for details. First stage statistics for the IV estimates are reported in Table 1. Point estimates and 95% confidence intervals.



Figure A18: Estimated cumulative effects

Notes: Estimates of equation 3 on INPS data, see Section 7.1 for details. First stage statistics for the IV estimates are reported in Table 1. Point estimates and 95% confidence intervals.



Figure A19: Estimated cumulative effects

Notes: Estimates of equation 3 on INPS data, see Section 7.1 for details. First stage statistics for the IV estimates are reported in Table 1. Point estimates and 95% confidence intervals.



Figure A20: Estimated cumulative effects

Notes: Estimates of equation 3 on INPS data, see Section 7.1 for details. First stage statistics for the IV estimates are reported in Table 1. Point estimates and 95% confidence intervals.



Figure A21: The wage-age gradient for all workers in the overall economy

Notes: Our calculations on INPS data. Data cover the whole private non agricultural sector; only employees working at least 150 days in a given year are included. Each wage level is normalized to the average for 40-44 years old in each year.

Authors	Data	Outcome	Old (y.o.)	Empirical Strategy	Instrumental variable	Effect
Daveri and Maliranta (2007)	Finland, electronics, forest, ind. machinery 1990-2002	Value added per hour worked, gross wage, TFP index	Dep. var.: age	FE, IV controls for experience and seniority	Lagged values for experience and seniority	Limited eff. of age on prod., larger on wage; seniority and exp matter more; prod/pay gap ↑ in electronics; no eff in traditional industries
Malmberg et al. (2008)	Sweden, manufac- turing and mining 1985-1996 Avg. firm size: 80	Value added per worker	50+	FE (dev. from the mean) IV (for avg. age)	Lagged values of age shares	Slightly positive eff on VA p.w. in larger firms (50+)
Martins et al. (2009)	Portugal non-farm priv. sector, 1992-1997 Avg. firm size:< 100	Hirings, separations, net job creation, of women, sales	57-60	DD+matching (age*time)	Pension reform stricter elig. requirements for women	Hirings and separations $\downarrow$ no eff. on net job creation negative eff on sales no in p.w. terms
Van Ours and Stoeldraijer (2011)	Netherlands, manufacturing 2000-2005 Avg. firm size: 97 (FTE: 77)	Value added, labor costs per FTE	50-56, 57+	FE, FD, GMM	Lagged values of age shares	Slightly positive eff. on prod. and wage no prod./pay gap
Cardoso et al. (2011)	Portugal, manuf. and service (priv. sect) 1986-2008 Empl(log): 3.2	Sales, labor costs p.w.	50-54, 55+	FD-IV-GMM	Lagged values of age shares	No negative eff. on sales and wage, no prod/ pay gap
Dostie (2011)	Canada, economy, no agric. 1999-2005 Avg. firm size: 15	Value added, avg. wage	55+	Two-stage estimation	Inverted demand funct for capital as proxy for prod. shock	No negative eff. on prod and wage, no prod/ pay gap Heterogeneity by gender (< 0 men with degree)

# Table A1: Literature review: estimates of the effects of aging workforce on firm outcomes

Authors	Data	Outcome	Old (y.o.)	Empirical Strategy	Instrumental variable	Effect
Cataldi et al. (2012)	Belgium, private sector 1999-2006 Avg. firm size: 100	Value added per hour worked, gross hourly wage	50+ (changes in work hours)	Pooled OLS, FD	Lagged values of age shares	Negative eff. on prod, positive on avg. wage (ref. group: prime-aged) prod./pay gap ↑ with age
Göbel and Zwick (2012)	Germany, manuf. and services 1997-2005 Avg. firm size: 176 (FTE)	Value added p.w. (sales-interm. inputs)	50-54, 55-60	Pooled OLS, dynamic GMM (controls for experience and seniority)	Lagged values of age shares	No negative eff. on prod. (ref. group: 35-40 y.o.) No sectoral diff
Mahlberg et al. (2013)	Austria, private sector (no agric.) 2002-2005 Avg. firm size: 70	Value added p.w., average wage	49+	OLS with lagged regressors, FE, RE, GMM	Lagged values of age shares	No effect on VA p.w. and avg. wage; negative eff. on avg. wage of the share of younger workers $(\leq 29)$ (ref. group: prime-aged)
Vandenberghe $(2013)^a$	Belgium, priv. sector 1998-2006 Empl(log): 3.9	Value added, labor costs, gross profits p.w.	50-64 men and women	FD-IV-GMM Two-stage estimation	Lagged values of age shares, interm. goods as proxy for prod. shock	Larger negative eff. on prod. than wage of older women, negative eff. on profits (ref. group: 30-49 men) Larger eff. in services and large firms
Ilmakunnas and Ilmakunnas (2015)	Finland, priv. non farm buss. sector 1994-1998	Hirings of old workers	49-57	DDD (age*size*time)	Pension reform stricter elig. requirements	Hirings increase mostly for 51-52 y.o., more strongly in larger firms

<sup>a</sup>The author claims that the observed increase in the share of older women is more likely to be exogenous than that in the share of older men since driven by the alignment of the legal retirement age of women to that of men. Thus, this provides a sort of "natural experiment".

Authors	Data	Outcome	Old (y.o.)	Empirical Strategy	Instrumental variable	Effect
Boeri et al. (2017)	Italy, priv. sector 2008-2014 15-150 empl.	Net empl. variation by age	55+	DD	Pension reform ind. and firm treatment	Negative effects on young workers (less than 30)
Bovini and Paradisi (2019)	Italy, priv. sector 2009-2015 3-200 empl.	Hirings and firings by age class	55+	DD	Pension reform ind. and firm treatment	Subst. between workers of different ages
Hut (2019)	Netherlands, priv. non fin sector 2001-2018 5+ empl.	Empl. by age profits, labour costs, inv.	54-57	DD (year of bith)	Pension reform ind. and firm treatment	Negative effects on financially constrained firms

Table A2:	Descriptive	statistics
-----------	-------------	------------

UNBALANCED PANEL	
Variable	
	_

Variable	Obs	Mean	Std. Dev.	Min	Max
Total employees	$9,\!654$	9654.0	608.9	3732.4	20.0
Employment share $(15-34, p.c.)$	$9,\!654$	23.4	13.7	0	90.6
Employment share (35-54, p.c.)	$9,\!654$	65.0	11.6	9.4	100
Employment share $(55+, p.c.)$	$9,\!654$	11.6	7.5	0	79.3
Net employment variation (15-34, p.c.)	$9,\!654$	1.1	5.2	-36.3	306.5
Net employment variation (35-54, p.c.)	$9,\!654$	0.0	9.0	-100.0	420.3
Net employment variation $(55+, p.c.)$	$9,\!654$	-1.1	2.3	-28.7	50.8
FTE Wage $(15-34, euros)$	$9,\!616$	87.8	18.2	53.7	148.5
FTE Wage $(35-54, euros)$	$9,\!654$	110.1	29.6	58.5	216.5
FTE Wage $(55+, euros)$	9,579	132.2	51.3	54.3	338.2
Wage var $(15-34, p.c.)$	$9,\!609$	2.0	5.7	-15.8	20.2
Wage var $(35-54, p.c.)$	$9,\!654$	1.7	4.5	-11.7	17.0
Wage var $(55+, p.c.)$	9,538	2.2	11.7	-28.7	47.0
Capital (euros*1000)	$9,\!654$	113691.1	1261317	3	57200000
Capital var. (p.c.)	$9,\!579$	-0.8	11.7	-22.7	25.6
Total labor costs ( $euro*1000$ )	$9,\!654$	18435.4	100517.6	20.9	3507677.0
Total labor costs var. (p.c.)	$9,\!579$	2.2	7.7	-10.4	21.1
Value added (euro*1000)	$9,\!654$	44863.7	314858.9	-453461.0	12000000.0
Value added var (p.c.)	$9,\!579$	1.7	16.5	-27.3	30.8

#### **BALANCED PANEL**

Variable	Obs	Mean	Std. Dev.	Min	Max
Total employees	$5,\!125$	691.2092	4678.207	21	140687
Employment share $(15-34, p.c.)$	$5,\!125$	22.9	13.3	0	90.6
Employment share $(35-54, p.c.)$	$5,\!125$	65.3	11.3	9.4	98.1
Employment share $(55+, p.c.)$	$5,\!125$	11.8	7.5	0	79.3
Net employment variation (15-34, p.c.)	$5,\!125$	1.0	3.7	-36.3	66.7
Net employment variation (35-54, p.c.)	$5,\!125$	0.1	5.8	-63.6	90.4
Net employment variation $(55+, p.c.)$	$5,\!125$	-1.1	2.0	-24.5	25.5
FTE Wage $(15-34, euros)$	$5,\!116$	88.24434	18.05651	53.65068	148.462
FTE Wage $(35-54, euros)$	$5,\!125$	110.4067	29.14231	58.53726	216.4977
FTE Wage $(55+, euros)$	5,102	133.2559	51.27009	54.32445	338.2433
Wage var (15-34, p.c.)	$5,\!113$	2.1	5.6	-15.8	20.2
Wage var (35-54, p.c.)	$5,\!125$	1.8	4.3	-11.7	17.0
Wage var $(55+, p.c.)$	5,088	2.3	11.4	-28.7	47.0
Capital (euros*1000)	$5,\!125$	82330.73	418256.8	25	9053672
Capital var. (p.c.)	$5,\!116$	-0.4	11.4	-22.7	25.6
Total labor costs ( $euro*1000$ )	$5,\!125$	20679.73	119983.3	407.049	3507677
Total labor costs var. (p.c.)	$5,\!116$	2.3	7.3	-10.4	21.1
Value added (euro*1000)	$5,\!125$	46594.1	261139	-453461	7683976
Value added var (p.c.)	$5,\!116$	2.3	16.1	-27.3	30.8

Notes: INVIND-INPS-CEBI data. Years: 2010-2014, selected sample: firms that employed at least 50 employees in the year they first appeared in the sample; firms which are sampled in each of the five years and do not have missing values on variables such as capital, labor costs and value added.

Table A3: Requirements for old age pension eligibility; changes in rules according to the law in place at time t and the law known at t - 1 for time t.

	L	$aw_{t t}$	La	$w_{t t-1}$	$Law_{t }$	$t - Law_{t t-1}$
	Men	Women	Men	Women	Men	Women
Year $t$						
2009	65	60	65	60	0	0
2010	65	60	65	60	0	0
2011	65	60	65	60	0	0
2012	66	62	65	60	1	2
2013	66.3	62.3	66.3	62.3	0	0
2014	66.3	63.8	66.3	63.8	0	0
2015	66.3	63.8	66.3	63.8	0	0
2016	66.6	65.6	66.6	65.6	0	0
2017	66.6	65.6	66.6	65.6	0	0

*Notes:* Old age pension eligibility requires the legal retirement age (reported in the Table) and at least 20 accrued years of social security contribution. We incorporate the 1-year waiting window (the number of months between retirement eligibility and actual pension disbursement) in the provision of statutory retirement age.

Table A4: Requirements for seniority pension eligibility; changes in rules according to the law in place at time t and the law known at t - 1 for time t.

		$Law_{t t}$		j	$Law_{t t-}$	1	Law	t - La	$w_{t t-1}$
	Quota	on	nly $C$	Quota	or	nly $C$	Quota	or	$\operatorname{ly} C$
	A, Q,	Men	Women	A, Q,	Men	Women	A	Men	Women
	$C \geq 35$			$C \geq 35$					
Year $t$									
2009	59,  95	40	40	59,  95	40	40	0	0	0
2010	59, 95	40	40	59, 95	40	40	0	0	0
2011	60, 96	40	40	60, 96	40	40	0	0	0
2012	-	42.1	41.1	60, 96	40	40	-	2.1	1.1
2013	-	42.4	41.4	-	42.4	41.4	-	0	0
2014	-	42.5	41.5	-	42.5	41.5	-	0	0
2015	-	42.5	41.5	-	42.5	41.5	-	0	0
2016	-	42.8	41.8	-	42.8	41.8	-	0	0
2017	-	42.8	41.8	-	42.8	41.8	-	0	0

Notes: A stands for age, C for number of years of social security contribution. Quota= A + C is the sum of age and years of social security contribution, which must be larger or equal than Q to reach retirement eligibility. Also requirements on minimum retirement age (A) and accrued years of contribution (C) are binding. Alternatively, retirement eligibility is also granted when the number of accrued years of social security contribution is higher than a minimum amount (i.e. 39 in 2007, 40 in 2008). We incorporate the 1-year waiting window (the number of months between retirement eligibility and actual pension disbursement) in the provision of statutory retirement age.

		Wages	
VARIABLES	15-34	35-54	55 +
FD OLS	-0.126	0.175	-1.507*
Net empl var $55+$	(0.164)	(0.387)	(0.859)
	5113	5125	5088
Obs.			
FD OLS with firm FE			
Net empl var $55+$	-0.109	0.301	-2.203**
	(0.190)	(0.449)	(0.912)
Obs.	5113	5125	5088
FD (Years 2011-12)			
	-0.0651	0.511	-3.044*
	(0.450)	(0.827)	(1.764)
	1022	1025	1020
FD IV (Years 2011-12)			
Net empl var 55+	-2.903	-3.345	-14.36
	(1.998)	(2.814)	(10.28)
Obs.	1022	1025	1020

Table A5: Wages var net of composition effects

*Notes:* Estimates of equation  $\overline{3}$ . All regressions include sector and year fixed effects and the 2010 shares of 35-54 and 55+ workers. Standard errors in brackets clustered at the firm level.

\* significant at 10%; \*\* significant at 5%; \*\*\* significant at 1%.

Column	1	2
VARIABLES	Exit	Entry
FD OLS		
Net empl var $55+$	-0.034	0.211
	(0.026)	(0.156)
FD OLS with FE		
Net empl var $55+$	-0.0196	0.00915
	(0.020)	(0.144)
Obs	13.176	13.176

#### Table A6: Check for attrition: sample entry and exit

*Notes:* Estimates of equation 3. All regressions include sector and year fixed effects and the 2010 shares of 35-54 and 55+ workers. Standard errors in brackets clustered at the firm level.

	15-34	35 - 54	55 +
FD OLS	0.078*	0.044	$0.098^{**}$
Net empl var $55+$	(0.043)	(0.045)	(0.048)
	5113	5125	5088
Obs			
FD OLS with FE	0.107**	0.046	$0.119^{**}$
Net empl var $55+$	(0.050)	(0.047)	(0.059)
	5113	5125	5088
Obs			
FD OLS (Years 2011-12)	-0.021	-0.049	$0.183^{**}$
	(0.105)	(0.034)	(0.081)
	1022	1025	1020
Obs			
FD IV (Years 2011-12)	-0.225	0.080	$1.451^{*}$
Net empl var 55+	(0.743)	(0.248)	(0.835)
		. ,	. /
Obs	1022	1025	1020

## Table A7: Variation in part time shares

*Notes:* Estimates of equation 3. All regressions include sector and year fixed effects and the 2010 shares of 35-54 and 55+ workers. Standard errors in brackets clustered at the firm level.

years 2012-11
dataset:
INPS
alternative
an
on
obtained
results
Main
Table A8:

Unbalanced sample All firms Firms less than 50 empl Firms with 50 or more empl	o cumpa Obs	65899	61879	65899	64691	55348	50499	65337	50683	50592	- i i
	Se	(0.145)	(0.001)	(0.433)	(0.006)	(0.007)	(0.023)	(0.017)	(0.015)	(0.012)	
	Coef	$1.132^{***}$	$-0.018^{***}$	$2.911^{**}$	-0.010	$-0.032^{***}$	0.004	$0.047^{**}$	$0.075^{***}$	$0.058^{***}$	
	Obs	1423957	875306	1423957	959191	299619	360730	1330496	365660	465220	
	Se	(0.093)	(0.003)	(0.105)	(0.003)	(0.006)	(0.012)	(0.027)	(0.0122)	(0.00866)	+ 1 +
	Coef	$0.747^{***}$	$-0.023^{***}$	$1.065^{***}$	-0.038***	$0.025^{***}$	$0.143^{***}$	$0.644^{***}$	$0.470^{***}$	$0.277^{**}$	J i 1:
	Obs	1489856	937185	1489856	1023882	354967	411229	1395833	416343	415812	0
	Se	(0.069)	(0.004)	(0.115)	(0.007)	(0.006)	(0.017)	(0.073)	(0.035)	(0.021)	0
	Coef	$0.943^{***}$	$-0.021^{***}$	$2.005^{***}$	$-0.020^{***}$	0.003	$0.070^{***}$	$0.347^{***}$	$0.242^{***}$	$0.152^{***}$	
		15-34	15-34	35-54	35-54	55+		INPS	Balance sheet		
		Net empl var	FTE wage var	Net empl var	FTE wage var	FTE wage var	Capital	Labor costs	Labor costs	Value added	T-4

Notes: Estimates of equation 3 on the who paid Social Security Contributions for at least one day in the relevant year. Regressions include as additional controls: the share of 35-54 and 55+ workers in 2010, sector and year fixed effects. Standard errors in brackets clustered at

the firm level.

#### RECENTLY PUBLISHED "TEMI" (\*)

- N. 1271 An analysis of sovereign credit risk premia in the euro area: are they explained by local or global factors?, by Sara Cecchetti (March 2020).
- N. 1272 Mutual funds' performance: the role of distribution networks and bank affiliation, by Giorgio Albareto, Andrea Cardillo, Andrea Hamaui and Giuseppe Marinelli (April 2020).
- N. 1273 Immigration and the fear of unemployment: evidence from individual perceptions in Italy, by Eleonora Porreca and Alfonso Rosolia (April 2020).
- N. 1274 Bridge Proxy-SVAR: estimating the macroeconomic effects of shocks identified at high-frequency, by Andrea Gazzani and Alejandro Vicondoa (April 2020).
- N. 1275 Monetary policy gradualism and the nonlinear effects of monetary shocks, by Luca Metelli, Filippo Natoli and Luca Rossi (April 2020).
- N. 1276 Spend today or spend tomorrow? The role of inflation expectations in consumer behaviour, by Concetta Rondinelli and Roberta Zizza (April 2020).
- N. 1277 Going the extra mile: effort by workers and job-seekers, by Matthias S. Hertweck, Vivien Lewis and Stefania Villa (June 2020).
- N. 1278 *Trainspotting: board appointments in private firms*, by Audinga Baltrunaite and Egle Karmaziene (June 2020).
- N. 1279 *The role of bank supply in the Italian credit market: evidence from a new regional survey*, by Andrea Orame (June 2020).
- N. 1280 The non-linear effects of the Fed asset purchases, by Alessio Anzuini (June 2020).
- N. 1281 The effects of shop opening hours deregulation: evidence from Italy, by Lucia Rizzica, Giacomo Roma and Gabriele Rovigatti (June 2020).
- N. 1282 *How do house prices respond to mortgage supply*?, by Guglielmo Barone, Francesco David, Guido de Blasio and Sauro Mocetti (June 2020).
- N. 1283 The macroeconomics of hedging income shares, by Adriana Grasso, Juan Passadore and Facundo Piguillem (June 2020).
- N. 1284 Uncertainty matters: evidence from a high-frequency identification strategy, by Piergiorgio Alessandri, Andrea Gazzani and Alejandro Vicondoa (June 2020).
- N. 1285 *The economic drivers of volatility and uncertainty*, by Andrea Carriero, Francesco Corsello and Massimiliano Marcellino (July 2020).
- N. 1286 *Protectionism and the effective lower bound in the euro area*, by Pietro Cova, Alessandro Notarpietro and Massimiliano Pisani (July 2020).
- N. 1287 *Court efficiency and aggregate productivity: the credit channel*, by Guzmán González-Torres and Giacomo Rodano (July 2020).
- N. 1288 *The time-varying risk of Italian GDP*, by Fabio Busetti, Michele Caivano, Davide Delle Monache and Claudia Pacella (July 2020).
- N. 1289 *Trade shocks and credit reallocation*, by Stefano Federico, Fadi Hassan and Veronica Rappoport (July 2020).
- N. 1290 The impact of the IRB approach on the relationship between the cost of credit for public companies and financial market conditions, by Raffaele Gallo (July 2020).

<sup>(\*)</sup> Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

2018

- ACCETTURO A., V. DI GIACINTO, G. MICUCCI and M. PAGNINI, Geography, productivity and trade: does selection explain why some locations are more productive than others?, Journal of Regional Science, v. 58, 5, pp. 949-979, WP 910 (April 2013).
- ADAMOPOULOU A. and E. KAYA, *Young adults living with their parents and the influence of peers*, Oxford Bulletin of Economics and Statistics, v. 80, pp. 689-713, WP 1038 (November 2015).
- ANDINI M., E. CIANI, G. DE BLASIO, A. D'IGNAZIO and V. SILVESTRINI, *Targeting with machine learning:* an application to a tax rebate program in Italy, Journal of Economic Behavior & Organization, v. 156, pp. 86-102, WP 1158 (December 2017).
- BARONE G., G. DE BLASIO and S. MOCETTI, *The real effects of credit crunch in the great recession: evidence from Italian provinces*, Regional Science and Urban Economics, v. 70, pp. 352-59, **WP 1057 (March 2016)**.
- BELOTTI F. and G. ILARDI Consistent inference in fixed-effects stochastic frontier models, Journal of Econometrics, v. 202, 2, pp. 161-177, WP 1147 (October 2017).
- BERTON F., S. MOCETTI, A. PRESBITERO and M. RICHIARDI, *Banks, firms, and jobs,* Review of Financial Studies, v.31, 6, pp. 2113-2156, WP 1097 (February 2017).
- BOFONDI M., L. CARPINELLI and E. SETTE, *Credit supply during a sovereign debt crisis*, Journal of the European Economic Association, v.16, 3, pp. 696-729, WP 909 (April 2013).
- BOKAN N., A. GERALI, S. GOMES, P. JACQUINOT and M. PISANI, EAGLE-FLI: a macroeconomic model of banking and financial interdependence in the euro area, Economic Modelling, v. 69, C, pp. 249-280, WP 1064 (April 2016).
- BRILLI Y. and M. TONELLO, Does increasing compulsory education reduce or displace adolescent crime? New evidence from administrative and victimization data, CESifo Economic Studies, v. 64, 1, pp. 15–4, WP 1008 (April 2015).
- BUONO I. and S. FORMAI *The heterogeneous response of domestic sales and exports to bank credit shocks,* Journal of International Economics, v. 113, pp. 55-73, WP 1066 (March 2018).
- BURLON L., A. GERALI, A. NOTARPIETRO and M. PISANI, Non-standard monetary policy, asset prices and macroprudential policy in a monetary union, Journal of International Money and Finance, v. 88, pp. 25-53, WP 1089 (October 2016).
- CARTA F. and M. DE PHLIPPIS, You've Come a long way, baby. Husbands' commuting time and family labour supply, Regional Science and Urban Economics, v. 69, pp. 25-37, WP 1003 (March 2015).
- CARTA F. and L. RIZZICA, *Early kindergarten, maternal labor supply and children's outcomes: evidence from Italy*, Journal of Public Economics, v. 158, pp. 79-102, WP 1030 (October 2015).
- CASIRAGHI M., E. GAIOTTI, L. RODANO and A. SECCHI, A "Reverse Robin Hood"? The distributional implications of non-standard monetary policy for Italian households, Journal of International Money and Finance, v. 85, pp. 215-235, WP 1077 (July 2016).
- CIANI E. and C. DEIANA, *No Free lunch, buddy: housing transfers and informal care later in life*, Review of Economics of the Household, v.16, 4, pp. 971-1001, **WP 1117 (June 2017).**
- CIPRIANI M., A. GUARINO, G. GUAZZAROTTI, F. TAGLIATI and S. FISHER, *Informational contagion in the laboratory*, Review of Finance, v. 22, 3, pp. 877-904, WP 1063 (April 2016).
- DE BLASIO G, S. DE MITRI, S. D'IGNAZIO, P. FINALDI RUSSO and L. STOPPANI, *Public guarantees to SME borrowing*. A RDD evaluation, Journal of Banking & Finance, v. 96, pp. 73-86, WP 1111 (April 2017).
- GERALI A., A. LOCARNO, A. NOTARPIETRO and M. PISANI, *The sovereign crisis and Italy's potential output*, Journal of Policy Modeling, v. 40, 2, pp. 418-433, **WP 1010 (June 2015).**
- LIBERATI D., An estimated DSGE model with search and matching frictions in the credit market, International Journal of Monetary Economics and Finance (IJMEF), v. 11, 6, pp. 567-617, WP 986 (November 2014).
- LINARELLO A., Direct and indirect effects of trade liberalization: evidence from Chile, Journal of Development Economics, v. 134, pp. 160-175, WP 994 (December 2014).
- NATOLI F. and L. SIGALOTTI, *Tail co-movement in inflation expectations as an indicator of anchoring,* International Journal of Central Banking, v. 14, 1, pp. 35-71, WP 1025 (July 2015).
- NUCCI F. and M. RIGGI, *Labor force participation, wage rigidities, and inflation,* Journal of Macroeconomics, v. 55, 3 pp. 274-292, WP 1054 (March 2016).
- RIGON M. and F. ZANETTI, *Optimal monetary policy and fiscal policy interaction in a non\_ricardian economy,* International Journal of Central Banking, v. 14 3, pp. 389-436, WP 1155 (December 2017).

SEGURA A., Why did sponsor banks rescue their SIVs?, Review of Finance, v. 22, 2, pp. 661-697, WP 1100 (February 2017).

2019

- ALBANESE G., M. CIOFFI and P. TOMMASINO, *Legislators' behaviour and electoral rules: evidence from an Italian reform*, European Journal of Political Economy, v. 59, pp. 423-444, **WP 1135 (September 2017).**
- APRIGLIANO V., G. ARDIZZI and L. MONTEFORTE, Using the payment system data to forecast the economic activity, International Journal of Central Banking, v. 15, 4, pp. 55-80, WP 1098 (February 2017).
- ARNAUDO D., G. MICUCCI, M. RIGON and P. ROSSI, Should I stay or should I go? Firms' mobility across banks in the aftermath of the financial crisis, Italian Economic Journal / Rivista italiana degli economisti, v. 5, 1, pp. 17-37, WP 1086 (October 2016).
- BASSO G., F. D'AMURI and G. PERI, *Immigrants, labor market dynamics and adjustment to shocks in the euro area,* IMF Economic Review, v. 67, 3, pp. 528-572, WP 1195 (November 2018).
- BATINI N., G. MELINA and S. VILLA, *Fiscal buffers, private debt, and recession: the good, the bad and the ugly,* Journal of Macroeconomics, v. 62, WP 1186 (July 2018).
- BURLON L., A. NOTARPIETRO and M. PISANI, *Macroeconomic effects of an open-ended asset purchase programme*, Journal of Policy Modeling, v. 41, 6, pp. 1144-1159, **WP 1185 (July 2018).**
- BUSETTI F. and M. CAIVANO, Low frequency drivers of the real interest rate: empirical evidence for advanced economies, International Finance, v. 22, 2, pp. 171-185, WP 1132 (September 2017).
- CAPPELLETTI G., G. GUAZZAROTTI and P. TOMMASINO, *Tax deferral and mutual fund inflows: evidence from a quasi-natural experiment*, Fiscal Studies, v. 40, 2, pp. 211-237, **WP 938 (November 2013).**
- CARDANI R., A. PACCAGNINI and S. VILLA, *Forecasting with instabilities: an application to DSGE models* with financial frictions, Journal of Macroeconomics, v. 61, WP 1234 (September 2019).
- CHIADES P., L. GRECO, V. MENGOTTO, L. MORETTI and P. VALBONESI, Fiscal consolidation by intergovernmental transfers cuts? The unpleasant effect on expenditure arrears, Economic Modelling, v. 77, pp. 266-275, WP 985 (July 2016).
- CIANI E., F. DAVID and G. DE BLASIO, *Local responses to labor demand shocks: a re-assessment of the case of Italy*, Regional Science and Urban Economics, v. 75, pp. 1-21, WP 1112 (April 2017).
- CIANI E. and P. FISHER, *Dif-in-dif estimators of multiplicative treatment effects*, Journal of Econometric Methods, v. 8. 1, pp. 1-10, WP 985 (November 2014).
- CIAPANNA E. and M. TABOGA, *Bayesian analysis of coefficient instability in dynamic regressions*, Econometrics, MDPI, Open Access Journal, v. 7, 3, pp.1-32, WP 836 (November 2011).
- COLETTA M., R. DE BONIS and S. PIERMATTEI, *Household debt in OECD countries: the role of supply-side* and demand-side factors, Social Indicators Research, v. 143, 3, pp. 1185–1217, **WP 989 (November** 2014).
- COVA P., P. PAGANO and M. PISANI, *Domestic and international effects of the Eurosystem Expanded Asset Purchase Programme*, IMF Economic Review, v. 67, 2, pp. 315-348, **WP 1036 (October 2015).**
- ERCOLANI V. and J. VALLE E AZEVEDO, *How can the government spending multiplier be small at the zero lower bound?*, Macroeconomic Dynamics, v. 23, 8. pp. 3457-2482, **WP 1174 (April 2018).**
- FERRERO G., M. GROSS and S. NERI, *On secular stagnation and low interest rates: demography matters,* International Finance, v. 22, 3, pp. 262-278, **WP 1137 (September 2017).**
- FOA G., L. GAMBACORTA, L. GUISO and P. E. MISTRULLI, *The supply side of household finance*, Review of Financial Studies, v.32, 10, pp. 3762-3798, **WP 1044 (November 2015).**
- GIORDANO C., M. MARINUCCI and A. SILVESTRINI, *The macro determinants of firms' and households' investment: evidence from Italy*, Economic Modelling, v. 78, pp. 118-133, WP 1167 (March 2018).
- GOMELLINI M., D. PELLEGRINO and F. GIFFONI, *Human capital and urban growth in Italy*,1981-2001, Review of Urban & Regional Development Studies, v. 31, 2, pp. 77-101, **WP 1127 (July 2017).**
- MAGRI S., Are lenders using risk-based pricing in the Italian consumer loan market? The effect of the 2008 crisis, Journal of Credit Risk, v. 15, 1, pp. 27-65, WP 1164 (January 2018).
- MAKINEN T., A. MERCATANTI and A. SILVESTRINI, *The role of financial factors for european corporate investment*, Journal of International Money and Finance, v. 96, pp. 246-258, **WP 1148 (October 2017).**
- MIGLIETTA A., C. PICILLO and M. PIETRUNTI, *The impact of margin policies on the Italian repo market*, The North American Journal of Economics and Finance, v. 50, **WP 1028 (October 2015).**

- MONTEFORTE L. and V. RAPONI, Short-term forecasts of economic activity: are fortnightly factors useful?, Journal of Forecasting, v. 38, 3, pp. 207-221, WP 1177 (June 2018).
- NERI S. and A. NOTARPIETRO, Collateral constraints, the zero lower bound, and the debt-deflation mechanism, Economics Letters, v. 174, pp. 144-148, WP 1040 (November 2015).
- PEREDA FERNANDEZ S., *Teachers and cheaters. Just an anagram?*, Journal of Human Capital, v. 13, 4, pp. 635-669, WP 1047 (January 2016).
- RIGGI M., Capital destruction, jobless recoveries, and the discipline device role of unemployment, Macroeconomic Dynamics, v. 23, 2, pp. 590-624, WP 871 (July 2012).

2020

- BRIPI F., D. LOSCHIAVO and D. REVELLI, Services trade and credit frictions: evidence with matched bank firm data, The World Economy, v. 43, 5, pp. 1216-1252, WP 1110 (April 2017).
- COIBION O., Y. GORODNICHENKO and T. ROPELE, *Inflation expectations and firms' decisions: new causal evidence*, Quarterly Journal of Economics, v. 135, 1, pp. 165-219, WP 1219 (April 2019).
- CORSELLO F. and V. NISPI LANDI, *Labor market and financial shocks: a time-varying analysis*, Journal of Money, Credit and Banking, v. 52, 4, pp. 777-801, **WP 1179 (June 2018).**
- D'IGNAZIO A. and C. MENON, *The causal effect of credit Guarantees for SMEs: evidence from Italy,* The Scandinavian Journal of Economics, v. 122, 1, pp. 191-218, **WP 900 (February 2013).**
- RAINONE E. and F. VACIRCA, *Estimating the money market microstructure with negative and zero interest rates*, Quantitative Finance, v. 20, 2, pp. 207-234, WP 1059 (March 2016).
- RIZZICA L., Raising aspirations and higher education. Evidence from the UK's widening participation policy, Journal of Labor Economics, v. 38, 1, pp. 183-214, WP 1188 (September 2018).

#### FORTHCOMING

- ARDUINI T., E. PATACCHINI and E. RAINONE, *Treatment effects with heterogeneous externalities*, Journal of Business & Economic Statistics, **WP 974 (October 2014).**
- BALTRUNAITE A., C. GIORGIANTONIO, S. MOCETTI and T. ORLANDO, *Discretion and supplier selection in public procurement*, Journal of Law, Economics, and Organization, WP 1178 (June 2018).
- BOLOGNA P., A. MIGLIETTA and A. SEGURA, *Contagion in the CoCos market? A case study of two stress events*, International Journal of Central Banking, WP 1201 (November 2018).
- BOTTERO M., F. MEZZANOTTI and S. LENZU, *Sovereign debt exposure and the Bank Lending Channel: impact on credit supply and the real economy,* Journal of International Economics, **WP 1032 (October 2015).**
- BRONZINI R., G. CARAMELLINO and S. MAGRI, Venture capitalists at work: a Diff-in-Diff approach at latestages of the screening process, Journal of Business Venturing, WP 1131 (September 2017).
- BRONZINI R., S. MOCETTI and M. MONGARDINI, *The economic effects of big events: evidence from the Great Jubilee 2000 in Rome*, Journal of Regional Science, WP 1208 (February 2019).
- COVA P. and F. NATOLI, *The risk-taking channel of international financial flows*, Journal of International Money and Finance, **WP 1152 (December 2017).**
- COVA P., P. PAGANO, A. NOTARPIETRO and M. PISANI, Secular stagnation, R&D, public investment and monetary policy: a global-model perspective, Macroeconomic Dynamics, WP 1156 (December 2017).
- DEL PRETE S. and S. FEDERICO, *Do links between banks matter for bilateral trade? Evidence from financial crises*, Review of World Economics, WP 1217 (April 2019).
- GERALI A. and S. NERI, *Natural rates across the Atlantic*, Journal of Macroeconomics, WP 1140 (September 2017).
- LIBERATI D. and M. LOBERTO, *Taxation and housing markets with search frictions*, Journal of Housing Economics, WP 1105 (March 2017).
- LOSCHIAVO D., Household debt and income inequality: evidence from Italian survey data, Review of Income and Wealth, WP 1095 (January 2017).
- MOCETTI S., G. ROMA and E. RUBOLINO, *Knocking on parents' doors: regulation and intergenerational mobility*, Journal of Human Resources, WP 1182 (July 2018).

- NISPI LANDI V. and A. SCHIAVONE, *The effectiveness of capital controls*, Open Economies Review, **WP 1200** (November 2018).
- PANCRAZI R. and M. PIETRUNTI, *Natural expectations and home equity extraction*, Journal of Housing Economics, **WP 984 (November 2014).**
- PEREDA FERNANDEZ S., Copula-based random effects models for clustered data, Journal of Business & Economic Statistics, WP 1092 (January 2017).

RAINONE E., The network nature of otc interest rates, Journal of Financial Markets, WP 1022 (July 2015).

- SANTIONI, R., F. SCHIANTARELLI and P. STRAHAN, *Internal capital markets in times of crisis: the benefit of group affiliation*, Review of Finance, WP 1146 (October 2017).
- SCHIANTARELLI F., M. STACCHINI and P. STRAHAN, Bank Quality, judicial efficiency and loan repayment delays in Italy, Journal of Finance, WP 1072 (July 2016).